# Overlapping Soft Comparators for High-Speed and Accurate ADCs

Presented in Partial Fulfillment of the Requirements for the Degree of

MASTER OF SCIENCE

with a Major in

**Electrical Engineering** 

in the

College of Graduate Studies

University of Idaho

by

Abu Md. Zafar Sadeque

APRIL 2016

Major Professor Saied Hemati, Ph.D., P.Eng.

Committee Suat Ay, Ph.D., P.E. Herbert Hess, Ph.D., P.E.

Department Administrator MOHSEN GUIZANI, PH.D.

## Authorization to Submit Thesis

This thesis of Abu Md. Zafar Sadeque, submitted for the degree of Master of Science with a Major in Electrical Engineering and titled "Overlapping Soft Comparators for High-Speed and Accurate ADCs," has been reviewed in final form. Permission, as indicated by the signatures and dates below, is now granted to submit final copies to the College of Graduate Studies for approval.

Major Professor:	Saied Hemati, Ph.D., P.Eng.	Date
Committee Members:	Suat Ay, Ph.D., P.E.	Date
	Herbert Hess, Ph.D., P.E.	Date
Department		
Administrator:		
	Mohsen Guizani, Ph.D.	Date

#### Abstract

Comparators are one of the main building blocks in ADCs and there are quite a few techniques for designing high-quality Comparators. Nevertheless, when two inputs of a comparator are very close to each other, perfect device-matching is required to reduce offset. Noise can also cause meta-stability and degrades the precision, and it takes a very long time for the comparator to settle down, which adversely impacts the conversion time and throughput of the ADCs. Furthermore, designing a large number of high quality comparators is not affordable, which explains why high-resolution Flash ADCs are not common. To address the aforementioned problem, in this work a new comparator design is proposed, named as the Soft Comparator. In the proposed Soft Comparator, the settling time is exploited to generate multiple bits instead of only one bit. This allows reducing the number of comparators in designing high resolution Flash ADCs by a big factor. It is capable of alleviating meta-stability and offset problems by relying on adjacent comparators that overlap with each other. Ultimately, it can speed up the data conversion time as the longest settling time will not be relevant anymore.

#### Acknowledgements

I wish to thank my major professor Dr. Saied Hemati for providing me the opportunity to conduct my Masters research under his supervision, for guiding and supporting me throughout my graduate research work, for providing me the laboratory facilities, for giving technical advices and finally for having patience on my abilities. I would like to give a big thanks to post-doctoral researcher Ismail Cevik, Ph.D. for helping me to be efficient with Cadence Virtuoso schematic and layout design. Especially his technical advices regarding chip layout design helped me to successfully complete my research work on time.

I would also like to thank Ph.D. student Hossein Mani for helping me in Soft Comparator Noise Analysis.

Finally, I would like to thank my mother for her prayers, constant mental support and encouragement throughout my life.

### Dedication

This work is dedicated to my parents,

Md. Tajul Islam and Dilruba Islam, who have provided me constant support and encouragement during difficult periods of my graduate school and life.This work is also dedicated to all the freedom fighters of Liberation War in 1971, who sacrificed their lives for my beloved motherland Bangladesh.

# TABLE OF CONTENTS

	Authorization to Submit Thesis	i
	Abstract	i
	Acknowledgements	7
	Dedication	7
	TABLE OF CONTENTS	i
	List of Tables	i
	List of Figures	(
1	CHAPTER 1.       1.1       Motivation and Background       1.1         1.1       Motivation and Background       1.2         1.2       Contributions       1.2         1.3       Thesis Organization       2	[ [ 2 2
2	2.1       Comparator Voltage Transfer Characteristics       5         2.2       Comparator Static Characteristics       7         2.2.1       Voltage Gain       7         2.2.2       Input offset       7         2.2.3       Input resolution       7         2.2.4       Noise       10         2.2.5       Input Common Mode Voltage Range       10         2.2.6       Output High and Low States       10         2.3       Comparator Dynamic Characteristics       11         2.3.1       Slew Rate       11         2.3.2       Propagation delay       12         2.4       Comparator Types       12         2.4       Continuous Comparator       14         2.4.1       Continuous Comparator       14	573))))) [[2347
3	CHAPTER 3.       24         3.1 Analog-to-Digital Converter Applications       24         3.2 Design Study of Analog-to-Digital Converters (ADCs)       27         3.2.1 Different Technologies and Architectures       27         3.2.2 Resolution and Sampling Rate       26         3.2.3 Power Consumption and Area       31         3.2.4 DNL and INL       32         3.2.5 Signal-to-Noise and Distortion Ratio (SNDR)       35         3.2.6 ADC performance summary (2010-2014)       36	1177 21 353

	3.3	High Speed-High Resolution Analog-to-Digital Converters (ADCs) 45
		2.2.2 Why high resolution Flash ADCs are not common
		2.2.2 Successive Approximation Register (SAR) ADC Meta-stability
	2.4	Challenges in designing High Speed High Posselution ADCs
	3.4	Chanenges in designing right speed-right Resolution ADCs
4 Cн	IAPT	TER 4
•	4.1	Soft and Hard Decision Decoding
	4.2	Soft Comparator and Hard Comparator concept
	1.3	Soft Comparator Architecture
	т.)	4.3.1 Soft Comparator schematic diagram
		4.2.2 Dynamic Latched Comparator
		4.3.2 Dynamic Euclide Comparator
		4.3.5 NOR Gate and Duner Circuit
		4.3.4 Delay Elle Circuit
		4.3.5 Memory Clicuit
	4.4	How Soft comparator works
	4.5	Soft Comparator Noise Analysis
		4.5.1 Single Soft Comparator
		4.5.2 M Soft Comparators with same threshold
5 Сн	тарт	YER 5
j en	E 1	Circuit Lavout Plan
	5.1	Soft Comparator Lavout
	5.2	a Dynamic Comparator
		5.2.1 Dynamic Comparator
		5.2.2 AOK Gale and Duller Circuit
		5.2.3 Big Delay Cell and Delay Line Circuit
		5.2.4 Memory Circuit $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$
		5.2.5 Encoder Circuit
		5.2.6 Soft Comparator
	5.3	Metal Filling and Capacitors
	5.4	Pad Frame Creation
	5.5	Final Chip Layout
	5.6	Chip Evaluation
		5.6.1 Performance test with Temperature variation
		5.6.2 Corner Simulation
		5.6.3 Offset Calculation: Monte Carlo Simulation
		5.6.4 Performance Summary
( C -		
0 CH	IAPT	EKO
	0.1	ruture vvorк
B11	BLIO	OGRAPHY

TABLE 2.1	Design parameters for the Continuous Comparator	18
TABLE 2.2	Design parameters for the Latched Comparator	23
TABLE 3.1	Brief summary of ADCs designed in 2010 published in JSSC	40
TABLE 3.2	Brief summary of ADCs designed in 2011 published in JSSC	41
TABLE 3.3	Brief summary of ADCs designed in 2012 published in JSSC	42
TABLE 3.4	Brief summary of ADCs designed in 2013 published in JSSC	43
TABLE 3.5	Brief summary of ADCs designed in 2014 published in JSSC	44
TABLE 4.1	Design parameters for the Latched Comparator	58
TABLE 4.2	Design parameters for the Big Tristate NOT Gate	61
TABLE 4.3	Design parameters for the Tristate NOT Gate	64
TABLE 5.1	Dualmimcap Capacitor values	87
TABLE 5.2	Corner Definition used in Corner Simulation	94
TABLE 5.3	Corner Simulation Summary (without calibration)	96
TABLE 5.4	Corner Simulation Summary (after calibration)	96
TABLE 5.5	Monte Carlo Simulation Summary	98
TABLE 5.6	Performance Summary of the Soft Comparator	98

Fig. 2.1	Non-inverting Comparator [1]	5
Fig. 2.2	Voltage transfer characteristic (Ideal) [1]	6
Fig. 2.3	Voltage transfer characteristic (Non-ideal) [1]	6
FIG. 2.4	Voltage transfer characteristic (Non-ideal, zero reference) [1]	7
FIG. 2.5	Voltage Gain of a Comparator [2]	8
Fig. 2.6	Input Offset of a Comparator [2]	9
Fig. 2.7	Comparator Noise [2]	10
Fig. 2.8	Slew Rate of a Comparator [3]	12
Fig. 2.9	Propagation Delay of Comparator [3]	13
Fig. 2.10	A BiCMOS Continuous-time comparator [4]	14
Fig. 2.11	Continuous-time Comparator with two comparison part [5]	15
Fig. 2.12	Continuous-time comparator with two stages [6]	16
Fig. 2.13	Regenerated Continuous-time Comparator in 130nm technology	16
Fig. 2.14	Regenerated Comparator input and output signals	17
Fig. 2.15	A typical Dynamic Latched Comparator [7]	19
Fig. 2.16	A Regenerative Latched Comparator [8]	19
Fig. 2.17	A Fully Dynamic Latched Comparator with Higher Gain [9]	20
Fig. 2.18	Energy efficient Dynamic two stage Comparator[10]	21
Fig. 2.19	Comparator node voltages during comparison [10]	21
Fig. 2.20	Regenerated Dynamic Latched Comparator in 180nm technology.	22
FIG. 2.21	Comparator nodes AP and AN during comparison	22
Fig. 2.22	Comparator nodes OUTP and OUTN during comparison	23
Fig. 3.1	Analog to digital conversion through ADC	24
Fig. 3.2	ADC types and applications based on speed and resolution [11]	25
Fig. 3.3	Number of different ADC architectures published between 2010-14.	27
Fig. 3.4	Number of ADCs with different technologies (nm)	28
Fig. 3.5	Number of ADCs with different Architectures	28
Fig. 3.6	ADC Resolution (Bits) with different Technologies (nm)	29
Fig. 3.7	ADC Resolution (Bits) with different Architectures	29
Fig. 3.8	ADC Sampling RateResolution (Bits) with different Architectures.	30
Fig. 3.9	ADC Average Power (mW) with Technologies (nm)	31
Fig. 3.10	ADC Average Power (mW) with different Architectures	32
Fig. 3.11	ADC Average area $(mm^2)$ with Technologies $(nm)$ .	32
Fig. 3.12	ADC Average area $(mm^2)$ with different Architectures	33
Fig. 3.13	Average DNL values with Technologies (nm)	33
Fig. 3.14	Average INL values with Technologies (nm)	34
Fig. 3.15	Average DNL values with different Architectures	34
Fig. 3.16	Average INL values with different Architectures	35
Fig. 3.17	Average SNDR values with Technologies (nm)	35
Fig. 3.18	Average SNDR values with different Architectures	36

Fig. 3.19 Fig. 3.20	SNDR values against Energy per Nyquist sample [12] SNDR values against frequencies $(f_{in \ h \ f})$ [12]	37 37
FIG 2 21	State-of-the-art analog-to-digital converters (ADCs)	45
FIC 2.22	Typical 2-bit Flash ADC architecture [7]	42
$\begin{array}{c} \text{Fig. 5.22} \\ \text{Fig. 5.22} \end{array}$	Typical SAR ADC architecture [12]	47
11G. 3.23		40
Fig. 4.1	Hard decision decoding system setup [14]	51
Fig. 4.2	Hard decision decoding [14]	51
Fig. 4.3	Soft decision decoding system setup [14]	52
FIG. 4.4	Soft decision decoding [14].	52
Fig. 4.5	Soft Comparator basic block diagram	53
Fig. 4.6	Soft comparator schematic diagram	54
Fig. 4.7	Soft Comparator output with $INP = 610 \text{mV}$ and $INN = 600 \text{mV}$ .	54
FIG. 4.8	Dynamic Two-stage Latched Comparator	55
, Fig. 4.9	Comparator nodes OUTP and OUTN during comparison	56
FIG. 4.10	Simplified model of Comparator in Latch Mode [7]	56
FIG. 4.11	Linearized model of track-and-latch stage in Latch Mode [7]	57
FIG. 4.12	Output Settling Time with Input Common Mode Voltage Difference	58
FIG. 4.13	XOR gate designed with 4 two-input NAND gate.	59
FIG. 4.14	XOR Gate checks Comparator Output settling	50
FIG 4 15	Big Delay Cell circuit	60
FIG  4.19	Big Tristate NOT Gate circuit	61
FIG 4.17	Delay Line Circuit symbol	62
FIG 4.18	Delay Line Circuit schematic diagram	62
FIG 4 10	8 Delay Cells connected together	62
FIG 4.20	Delay Cell circuit	63
FIG 4.21	Tristate NOT Gate circuit	62
FIG 4.21	D-latch based Memory circuit	64
FIC 4.22	22 D-latch connected in parallel	65
FIC $4.23$	Schematic of the used D-latch circuit	66
FIC 4.24	22-to-F Encoder Circuit symbol	66
FIG. 4.25	Schematic of 22-to-5 Encoder circuit	67
FIG. 4.20	Schematic of 16 bit OR gate	68
FIG. $4.27$	Single Soft Comparator	00
FIG. 4.20	Decision making time for different common mode voltage difference	70
FIG. 4.29	in a single soft comparator	<b>-</b> 1
Erc ( ac	Drobability of correct detection for different interval longths	71
FIG. 4.30	Probability of confect detection for different interval lengths	73
FIG. 4.31	Frobability of each interval when fixed $v_{in}$ is applied in interval $[0, T]$	
Erc	[0,1]	74
FIG. 4.32	Two abcompations with normal distribution	75
гіс. 4.33 Гіс	Iwo observations with normal distribution.	76
гіс. 4.34	Effect of multiple observations for multiplication method. A fixed	
	$v_{in}$ is applied in interval $[0, 1]$	79
Fig. 5.1	Soft Comparator Layout configuration plan	80

Fig. 5.2	Dynamic Comparator Layout.	81
Fig. 5.3	XOR Gate Layout	81
Fig. 5.4	Buffer Circuit Layout	82
Fig. 5.5	Big Delay Cell Layout	82
Fig. 5.6	Delay Line Circuit (Top and Bottom portions are shown)	83
Fig. 5.7	D-Latch Circuit Layout	83
Fig. 5.8	Memory Circuit Layout (Top and Bottom portions are shown)	84
Fig. 5.9	32-to-5 Encoder Layout and 16-bit OR Gate Layout	85
Fig. 5.10	Soft Comparator Layout	86
Fig. 5.11	Metal filling in the chip area	87
FIG. 5.12	Dualmimcap Capacitors used in final chip layout	88
Fig. 5.13	Analog Pad, Digital Input Pad, and Digital Output Pad	88
Fig. 5.14	Vdd Pad, Vcc Pad, and GND Pad	89
Fig. 5.15	Pad Frame Pin Configuration.	90
Fig. 5.16	Final Chip Layout.	91
FIG. 5.17	Decimal Equivalent of Digital Output change with Temperature	92
Fig. 5.18	Decimal Equivalent of Digital Output after $V_{dd1}$ Calibration	93
Fig. 5.19	Soft Comparator Output for typtyp (tt) Corners	94
Fig. 5.20	Soft Comparator Output for fastslow (fs) Corners	94
FIG. 5.21	Soft Comparator Output for slowfast (sf) Corners	95
FIG. 5.22	Soft Comparator Output for slowslow (ss) Corners	95
FIG. 5.23	Soft Comparator Output for fastfast (ff) Corners	95
FIG. 5.24	Monte Carlo Simulation Output of the Soft Comparator	97
Fig. 5.25	Longest Transition at the $29^{th}$ sample point in the simulation	97

#### INTRODUCTION

A comparator is a circuit which detects whether a signal is larger or smaller than another signal, or larger than zero or not. It is the second most widely used electronic circuit after operational amplifier (OpAmp). The principle application of comparators lie in designing analog-to-digital converter (ADC) circuits. Among others, data transmission, switching power regulators, threshold & zero crossing detectors, relaxation oscillators, level shifters are some of the most important applications. Thus to meet different application requirements, comparators must meet different key specifications like offset, noise, speed, power, hysteresis to name a few. A good comparator should possess good decision making speed, low error, low power consumption, and be easy to design.

#### **1.1 MOTIVATION AND BACKGROUND**

Analog-to-digital converters (ADCs) have different important applications in communications, sensors, music technology, microcontrollers, software defined radios, medical fields, etc. Comparators are one of the main building blocks in analogto-digital converters (ADCs). A comparator compares two analog input signals, or compares a single analog signal to zero and produces a single bit output (hard comparators), which is one or zero. Comparators perform better when the difference between the inputs is large. This makes the comparator fast and reduces error and as a result ADCs can also be fast. There are some reasons behind this fact. First of all, the positive feedback builds up quickly that increases the comparator speed. Secondly, the small offset voltages due to device mismatch will not change the final decision and thirdly, the likelihood of meta-stability due to noise will be negligible. Unfortunately, in high resolution ADCs the voltage difference between the inputs is very small which is why the comparator needs a long settling time. It is a fundamental challenge to design a comparator which can be capable of comparing two very close signals without meta-stability, with good speed and accuracy. Also to build a high resolution Flash ADC, it is necessary to design a large number of accurate comparators, which is the reason behind difficulties in designing high resolution Flash ADCs.

#### **1.2 CONTRIBUTIONS**

There are different types of comparator circuits available. All these comparators compares two signals and produce only one single bit output – either 1 or 0. These comparators can be called *Hard Comparator* as they produce a hard decision output by telling whether the signal is larger or smaller than the other one. But comparator's decision making time is long and these circuits suffer from offset errors and metastabilities. Being slow in decision making, the comparators cannot be used in very high speed applications easily. Being affected by offset errors, their accuracy will be low and can generate unstable outputs because of meta-stability. To address these problems, a new comparator scheme has been proposed in this thesis. This new comparator circuit is designed with a latched comparator, some delay circuits, memory circuits, and an encoder module. In this design, the comparator settling time has been exploited to generate multiple bits per comparison, which is thus called a *Soft Comparator*. By using multiple overlapping soft comparators, reliable outputs can be generated. This also improves the accuracy and speed of ADCs (SAR, Flash, etc.). Especially for high resolution Flash ADCs, a large number of comparators are needed, which is the reason high resolution Flash ADCs are not common in general. For example, for a 5-bit Flash ADC,  $(2^5 - 1)$  traditional comparators will be needed. But using only one of our proposed Soft Comparators, up to 4 bits can be generated. Thus ideally by using only 4 of these Soft Comparators, a 5-bit Flash ADC can be designed, resulting in less power consumption, better accuracy, and high speed operation.

#### **1.3 THESIS ORGANIZATION**

The thesis is organized in six chapters. Chapter 1 introduces the applications of comparators and analog-to-digital converters. It describes the background and motivation of the research work and also the problems in designing comparators in general. It also briefly describes the research contributions of this thesis. At the end, it discusses the thesis organization.

Chapter 2 discusses comparator voltage transfer characteristics and comparator static characteristics, such as voltage gain, input offset, input resolution, noise, input common mode voltage range, and output high and low states. It also goes over comparator dynamic characteristics, such as propagation delay and slew rate. At the end, it goes through comparator classifications, continuous comparators, and dynamic latched comparators. It also reviews literature regarding different state-of-art comparator designs. This chapter also includes one continuous comparator and one dynamic latched comparator design regeneration and input-output simulations.

Chapter 3 starts with briefing the importance of analog to digital data conversion and why it is so challenging to work in analog domain. It describes various applications of analog-to-digital converters (ADCs) and relates different types of ADCs to their specific applications. Major portions of this chapter then reviews the literature by discussing various ADC designs published in Journal of Solid-State Circuit (JSSC) from 2010 to 2015. It studies and focuses ADC designs from different perspectives and summarizes their changes over time. It then discusses high speed and high resolution ADCs and their design challenges, then investigates the reason behind the lesser number of high resolution, high speed ADCs. At the end, it focuses on some specific design challenges of Successive Approximation Register (SAR) ADCs and Flash ADCs.

Chapter 4 explains the soft comparator architecture and working principle. It starts with explaining soft and hard decision decoding, the concept of hard comparator and soft comparator, and the relation between them. It overviews the designing of a soft comparator architecture and its different sub-modules including a dynamic latched comparator, delay line circuits, memory circuits, and encoder circuits and their working flow in different stages. Then it summarizes the working principle of soft comparators. At the end of this chapter, soft comparator noise analysis is briefly discussed. It also shows how using multiple soft comparators can improve the input voltage value detection, which is helpful for designing high speed and accurate

analog-to-digital converters (ADCs).

Chapter 5 shows the circuit layout design. It discusses the initial layout plan for the soft comparator chip, then breaks down into different sub-module layout designs. It covers the layout of dynamic comparator, XOR gate, buffer circuit, delay line circuit, memory circuit, and the encoder circuit. It then discusses the analog, digital, power, and ground pad layout design, pad frame creation, image bevel, metal filling, and usage of capacitors and their layout design. It also focuses on the chip evaluation and power, speed, and voltage of the designed chip.

Chapter 6 briefly summarizes the accomplishments and applications of the soft comparator design, some alternative soft comparator design approaches as the future ideas, and some design challenges while working on the soft comparator.

#### **COMPARATOR OVERVIEW**

A comparator compares two analog input signals - one is the reference signal and the other is the input signal. It compares them and produces a binary output - one or zero. Thus, a comparator acts as a single bit analog to digital converter (ADC). For an ideal comparator, it takes zero amount of time to compare and produce the output. But for all practical comparators, there is a transition time which is needed to compare the signals and produce the output. Thus, for every comparator, output voltage changes with different input signals and creates a comparator voltage transfer characteristic. This chapter covers characteristics related to comparators designed with open-loop operational amplifiers (Op-Amps).



FIGURE 2.1: Non-inverting Comparator [1]

#### 2.1 COMPARATOR VOLTAGE TRANSFER CHARACTERISTICS

A simple comparator can be designed with the help of an OpAmp with open loop configuration. If the open loop gain of the OpAmp is *A* then the comparator output equation will be:

$$V_O = A(V_+ - V_-)$$
(2.1)

Where  $V_+$  and  $V_-$  represent the non-inverting and inverting input of the OpAmp circuit and output  $V_O$  can be anything within the upper voltage limit  $V_{DD}$  and lower

voltage limit  $V_{EE}$ . In this case,  $V_+$  is the input analog signal  $V_{in}$  and  $V_-$  is the reference voltage signal  $V_{ref}$ . The voltage transfer characteristics represents how the output voltage  $V_O$  changes with the input voltage  $V_{in}$ .

If  $V_{in}$  is greater than  $V_{ref}$ , the output signal  $V_O$  should be equal to  $V_{DD}$ . On the other hand, if  $V_{in}$  is less than  $V_{ref}$ , the output signal  $V_O$  should be equal to  $V_{EE}$ . This is an ideal comparator case, where the assumption is that the comparator doesn't need any transition or settling time. Thus, it can instantly change levels without any delay.



FIGURE 2.2: Voltage transfer characteristic (Ideal) [1]

But a real comparator does not follow the ideal voltage transfer characteristics. Here, the comparator needs some transition or decision settling time to make a decision. If  $V_{in}$  is greater than  $V_{\delta+}$  then the output signal  $V_O$  will be equal to  $V_{DD}$ . On the other hand, if  $V_{in}$  is less than  $V_{\delta-}$  then the output signal  $V_O$  will be equal to  $V_{EE}$ . If the  $V_{in}$  is in between  $V_{\delta+}$  and  $V_{\delta-}$ , then the output signal can be anything between  $V_{DD}$  and  $V_{EE}$ .



FIGURE 2.3: Voltage transfer characteristic (Non-ideal) [1]

Generally, in many comparators, the analog input signal is compared against the ground level that is the zero reference voltage. In that case,  $V_{ref}$  will be 0 and the voltage transition will go through the zero reference voltage point. Fig. 2.4 is showing the case where  $V_{ref}$  = 0 and the comparator is non-ideal.



FIGURE 2.4: Voltage transfer characteristic (Non-ideal, zero reference) [1]

Thus, for a practical comparator ( $V_{\delta+} - V_{\delta-}$ ) is the linear range and  $V_{DD}$  and  $V_{EE}$  are the saturation ranges. Here,  $V_{\delta+}$  and  $V_{\delta-}$  are dependent on the open loop gain of the comparator A.

$$v_{\delta+} = \frac{V_{DD}}{A} \tag{2.2}$$

$$v_{\delta-} = \frac{V_{EE}}{A} \tag{2.3}$$

For a practical comparator open loop voltage gain, A = 200000 and  $V_{DD}$  = 10V and  $V_{EE}$  = -10V. Thus, the value of  $V_{\delta+}$  and  $v_{\delta-}$  are very small. In other words, the operational amplifier will be saturated very quickly.

#### 2.2 COMPARATOR STATIC CHARACTERISTICS

Comparators can have both static and dynamic characteristics. Static characteristics don't change with time where dynamic characteristics change with time progress. Also, static characteristics are mainly related to device sizes and parameters. On the other hand, dynamic characteristics are related to randomness of circuit. The main static characteristics of comparators are:

- 1. Voltage gain
- 2. Input offset
- 3. Input resolution
- 4. Noise
- 5. Input common mode voltage range
- 6. Output high and low states

#### 2.2.1 Voltage Gain

Voltage gain represents the overall device of the comparator. It is the ratio of the output voltage change to the input voltage change. Output voltage normally changes from a low saturation to a high saturation voltage level. Thus, if the low saturation level is  $V_{OL}$  and high saturation level is  $V_{OH}$ , and input voltage changes from  $V_{IL}$  to  $V_{IH}$  (or, vice versa) then voltage gain ( $A_v$ ) of the comparator would be

$$A_{v} = \frac{V_{OH} - V_{OL}}{V_{IH} - V_{IL}}$$
(2.4)



FIGURE 2.5: Voltage Gain of a Comparator [2]

Here,  $V_{IH}$  denotes the smallest input voltage for which the output rises to  $V_{OH}$  and  $V_{IL}$  denotes the largest amount of input voltage for which the output remains at  $V_{OL}$ . If the voltage gain is high for the comparator, then it can work well with small voltage changes. In other words, the resolution will also be better for that comparator.

#### 2.2.2 Input offset

Comparators can suffer from an input offset voltage issue. When both inputs of the comparators are equal, ideally, there should not be any high or low output states. But due to input offset, output can have either a high or low state even if both inputs are equal. The input offset of a comparator is the input at which the output is equally likely to be high or low. Noise determines how the output statistics vary as a function of input [7]. Input offset can be modeled or represented by a series voltage source applied at either of the comparator inputs.



FIGURE 2.6: Input Offset of a Comparator [2]

Thus, input offset voltage  $V_{OS}$  is the amount of voltage needed to apply at the input to make the output voltage level equal to  $\frac{V_{OH} + V_{OL}}{2}$  when both comparator inputs are equal. The source of input offset can be the transistor size mismatch. Mismatch in threshold voltage of the transistors and transistor trans-conductance parameters can be other sources. Input offset can degrade the comparator resolution.

#### 2.2.3 Input resolution

Comparator input resolution can be described as the amount of voltage change in the input of a comparator to make a state change at the output, that is from low to high state or vice versa. Thus, it is the smallest voltage difference between the input and the reference voltage responsible for binary digit change at the comparator output.

#### 2.2.4 Noise

Noise is an important issue when the comparator has high resolution. Random circuit noise can cause the output of a comparator to change from one logic state to the other, even when the comparator input is held constant. Hence, in order to measure the input-offset voltage in the presence of circuit noise, the input voltage is measured that results in the output state of the comparator being high or low with equal likelihood. The input-referred noise is then observed by changing the input around this value and observing how the output statistics vary as a function of the input voltage around this dc offset [7].



FIGURE 2.7: Comparator Noise [2]

#### 2.2.5 Input Common Mode Voltage Range

Input common mode voltage range is the voltage range within which the comparator performs the comparison without any problem. It is abbreviated as ICMR. Within ICMR, all the transistors in the comparator will be in saturation. Thus, to ensure the normal operation of comparator, ICMR is very important. ICMR can be affected by noise voltage which can hamper the differential performance of the comparators.

#### 2.2.6 *Output High and Low States*

The output high and low states are also important static characteristics. The output high state  $V_{OH}$  is the amount of voltage which can represent binary level '1' and output low state  $V_{OL}$  is the voltage level which represents the output binary level 'o'.

Normally, output can have only these two states as the state transition time is very short for general comparators.

#### 2.3 COMPARATOR DYNAMIC CHARACTERISTICS

Comparators also possess dynamic characteristics. Some of the characteristics of comparators change with time advancement. Due to their time varying nature, they are considered as dynamic characteristics. The major dynamic characteristics of comparators are:

- 1. Slew rate
- 2. Propagation delay

#### 2.3.1 Slew Rate

If a pulse signal is applied to one input of a comparator to compare with the other input signal or reference signal, then due to the internal compensation capacitor and depending on the frequency of the input pulse signal, the comparator might not follow the signal properly. Thus, the ability of any comparator to change the output voltage with time can be defined as slew rate of that comparator. The slew rate comes from the capacitor voltage and current relationship as below,

$$i = C \frac{dV}{dt} \tag{2.5}$$

Where i is the capacitor current and V is the voltage across the capacitor. If the rising and/or, falling voltage of comparator becomes high very large, the comparator output signal might be limited by slew rate. From this current-voltage relationship, if current becomes limited, then voltage also becomes limited and deteriorates the signal.



FIGURE 2.8: Slew Rate of a Comparator [3]

Comparator rising edge slew rate and falling edge slew rate are different. From the figure, rising and falling edge slew rate are expressed as,

$$SR_{+} = \frac{\Delta V}{\Delta t_{r}} \tag{2.6}$$

$$SR_{-} = \frac{\Delta V}{\Delta t_f} \tag{2.7}$$

(2.8)

Where  $\Delta t_r$  and  $\Delta t_f$  are measured for 10% and 90% of maximum voltage swing and  $\Delta V$  is the voltage difference between output high voltage state and output low voltage state.

#### 2.3.2 Propagation delay

Propagation delay is one of the most important parameter of any comparator. It directly relates to the input signal frequency range. When a comparator compares two input signals, it needs some time to make the comparison decision. This time is the propagation delay. In a more specific way, it can be said, when the input analog signal crosses the reference signal, to the time when the comparator makes a decision and changes the output states is expressed as propagation delay. Propagation delay can happen in rising edge or falling edge of the signal. Thus, the overall propagation delay is calculated by averaging the rising edge delay and falling edge delay.

Propagation delay time in fact varies with the input analog signal amplitude, which is also related to the slew rate of the comparator. Thus, propagation delay is indirectly related to the slew rate. So for a slew rate limited comparator, the



FIGURE 2.9: Propagation Delay of Comparator [3]

propagation delay  $t_p$  can be expressed as,

$$t_p = \Delta T = \frac{\Delta V}{SR} = \frac{(V_{OH} - V_{OL})}{2 \cdot SR}$$
(2.9)

Where slew rate SR is,

$$SR = \frac{dV_O}{dt} = \frac{\Delta V}{t_p}$$
(2.10)

#### 2.4 COMPARATOR TYPES

There are various types of comparator circuits. Each comparator circuit has advantages and disadvantages of its own. Based on different types of applications and requirements, comparators are designed differently. But in general, almost all comparators can be categorized into two basic types –

- 1. Continuous Comparator
- 2. Clocked, Dynamic, or Latched Comparator

#### 2.4.1 *Continuous Comparator*

Continuous comparators are very fast. Normally, they can have output logic states either at '1' or 'o' all the time, given that there is higher or lower analog input signal compared to the reference signal. As this comparator has continuous operation, it can change the output state immediately with the change of the input signal. But as it is working continuously, it is also consuming power all the time. Thus, it is a power hungry and energy inefficient comparator type.

Continuous comparators can be built using both CMOS and BiCMOS circuits. One design comprises of BiCMOS which is of high speed and also of high precision is shown in Fig. 2.10. For this design, offset voltage is compensated sacrificing the total timing of comparison.



FIGURE 2.10: A BiCMOS Continuous-time comparator [4]

Here, both inputs of the differential amplifier are disconnected, input mismatched are stored in a capacitor and then this offset is subtracted from the original measurement. The inputs can be tied together and then output can be measured, which gives the offset measurement. The offset can then be subtracted from the original output measurement during normal operation of the comparator. Fig. 2.11 shows another design comprised of two comparators, which are connected together to form a comparator [5]. Here, the input stage is designed by 3 PMOS transistors and they are connected through diode connections.



FIGURE 2.11: Continuous-time Comparator with two comparison part [5]

Upper stage and lower stage comparators are totally different in terms of design as the lower stage also comprises a S-R latch for better gain and rail to rail performance. Only one differential input comes to the upper stage and compared against the reference voltage to detect the trigger level and the other input is idle at that time. Fig. 2.12 shows a continuous time comparator design [6], which consists of a differential amplifier followed by a common source amplifier stage. This comparator is used in designing a 9-bit pulse position modulation based ADC design.

A ramp signal is applied as a reference signal to one of the comparator's inputs. It takes the input signal in the other input of the comparator and compares the signal amplitude with the reference ramp signal amplitude. Whenever the input signal becomes smaller than the ramp signal, *stop* signal becomes high and remains high until it is smaller than the ramp signal again. This *stop* signal is then converted to a digital bit flow using the subsequent time-to-digital circuit converter.



FIGURE 2.12: Continuous-time comparator with two stages [6]



FIGURE 2.13: Regenerated Continuous-time Comparator in 130nm technology

In this thesis, the comparator in [6] was regenerated using 130nm CMOS technology with some design modifications and a 600mV supply voltage (Fig. 2.13). The regenerated comparator worked properly and produced the same output *stop* signal (Fig. 2.14). One ramp generator circuit was designed, which took the clock signal as input and produced ramp of regular intervals. Also, one current generator circuit was designed which supplied a fixed DC current for this continuous comparator circuit and generated the biasing current.



FIGURE 2.14: Regenerated Comparator input and output signals

#### 2.4.2 Latched Comparator

Continuous comparators keep comparing two analog input signals all the time even when the comparison is actually not needed. Thus, it can consume unwanted power, which makes it energy inefficient. Many applications including analog-to-digital converters (ADCs) and digital memory circuit needs the comparator to compare the input signals at certain points of operation. When the comparison is done at certain intervals, greater comparison accuracy is achieved and also power management improves and results in an energy efficient comparator circuit. Thus, latched comparators need *clock signal* — with the high state of the clock, *Comparison* is done and with low state, *Reset* operation is done which eventually removes the previous comparison results stored in the comparator. This is the reason why it is also called a *Clocked Comparator*. The comparison action depends on the state of the clock and as soon as the clock changes its state, the comparator also acts. Thus, this type of comparator is also known as a *Dynamic Comparator*.

High speed dynamic latched comparators normally consist of two major stages – *Pre-amplifier* stage and *Track-and-Latch* stage [7]. The pre-amplifier stage amplifies the input analog signal thus, it gains higher resolution than the original signal. The output of the pre-amplifier is still smaller than the digital part of the comparator circuit. This also helps to minimize the kickback effect drastically. The outputs of the pre-amplifier stage go to the track and latch stage. This stage amplifies the pre-

Transistor	(W/L) Ratio
T21	1µ/2µ
T22	8µ/2µ
T23	10µ/1µ
T24	10µ/1µ
T20	2.4µ/2.4µ
T25	3µ/1µ
T26	3µ/1µ
T27	4µ/2µ
T31	2.8µ/1µ
T16	240n/140n
T18	240n/140n
T15	240n/140n
T17	240n/140n

TABLE 2.1: Design parameters for the Continuous Comparator

amplifier output both in track phase and latch phase. A positive feedback is normally enabled during the latch phase. This positive feedback ultimately generates the digital signal at the output of the comparator.

Fig. 2.16 shows a regenerative comparator design, which consists of a pre-amplifier and a latch circuit with two modes of operation — latch mode and trach mode [8]. The pre-amplifier consists of two p-type diode connected MOSFETs acting as an active load for two n-type MOSFET which are biased by a fixed tail DC current. The latch circuit is made of a cross-coupled pair of inverters. In the track mode, the difference between input signals will be amplified by the pre-amplifier. At that time, the latch circuit will be disabled. In the latch mode, latch circuit is enabled and it takes the pre-amplifier output to regeneratively amplify the difference in order to get a rail to rail voltage which in turn creates the digital output states — either a high '1' or low 'o'. One advantage of using a regenerative latched comparator is that they are pretty much noise free when the comparison decision is made as the outputs will be totally disconnected from the latch circuit inputs. But this kind of latch circuit can face meta-stability issues during sampling and amplifying the voltage difference received from the pre-amplifier stage. Regeneration time which is defined as the time needed for creating a decision in logic level from the start of sampling time, is a function of



FIGURE 2.15: A typical Dynamic Latched Comparator [7]



FIGURE 2.16: A Regenerative Latched Comparator [8]

initial voltage differences between the inputs and also the latch time constant which also depends on the parasitic capacitance of the circuit.

Dynamic latched comparators are fast, generally low power consuming with full railto-rail swing and high gain. Thus, this kind of comparator has many applications, including high speed and low power analog-to-digital converter (ADC), digital memory circuit, sense amplifier, etc. But due to parasitic capacitance output load mismatches and offset issues, they can have poor accuracy. Pre-amplifiers are normally used to minimize the offset.



FIGURE 2.17: A Fully Dynamic Latched Comparator with Higher Gain [9]

The design in [9] introduces two inverters between the pre-amplifier and latch stage. With this extra circuitry, the gain of the latch stage improves significantly. Fig. 2.18 shows an energy efficient dynamic two stage comparator used in a SAR ADC [10]. The first stage is the pre-amplifier and the second stage acts as another voltage amplifier and also as a positive feedback amplifier, which helps to create rail-to-rail digital output voltage. During the low state of the clock signal, FP and FN nodes parasitic capacitors will be charged. During the high state of the clock, these charged nodes will start to discharge through two n-type MOSFET to ground. Thus, the voltages of the nodes FP and FN will reduce to *gnd* from supply voltage level. As the FN and FP nodes lose their charges, in the second stage, two p-type MOSFET start to switch ON and the SN and SP node voltages will start to rise. Depending on the INP and INN values, either of SN or SP will reach to high voltage and the other one will eventually be grounded, thus, creating rail-to-rail digital voltage at the output [10].

The node voltages of the comparator, FN and FP of the first stage and SN and SP of the second stage, are also shown in Fig. 2.19. The FN and FP nodes discharge from



FIGURE 2.18: Energy efficient Dynamic two stage Comparator[10]

 $V_b$  supply voltage level to *gnd* level while SP and SN nodes started from the *gnd* level to reach rail-to-rail output voltage  $V_b$  and *gnd*. This design is very energy efficient. When nodes FN and FP are completely discharged, the power dissipation in the first stage stops and when the output nodes SN and SP reach a rail-to-rail voltage level, power dissipation stops in the seconds stage. Thus, when the comparator is not active or not doing any comparison, the total power dissipation is zero. Comparably, the first stage is more power efficient than the second stage, and it also amplifies more than the second stage.



FIGURE 2.19: Comparator node voltages during comparison [10]

In this thesis, the comparator circuit in [10] is regenerated using IBM 180nm CMOS technology with supply voltage of 1V and common mode voltage of 500mV (Fig. 2.20). The regenerated comparator produced the same node voltage characteristics.



FIGURE 2.20: Regenerated Dynamic Latched Comparator in 180nm technology

Here, AN and AP node voltages represent the FN and FP nodes respectively, which show similar voltage discharge behavior (Fig. 2.21).



FIGURE 2.21: Comparator nodes AP and AN during comparison

The output nodes of the second stage OUTP and OUTN represent the same nodes SP and SN, respectively and behave in the same way. Here, input voltages are INNP = 850mV and INN = 500mV and the rail-to-rail output voltage is reached in less than

2ns.

Comparator is one of the main building block of designing an analog-to-digital



FIGURE 2.22: Comparator nodes OUTP and OUTN during comparison

Transistor	(W/L) Ratio
TP6	2.4µ/200n
TN <sub>7</sub>	1.2µ/200n
ТРо	1.2µ/200n
TP1	1.2µ/200n
TNo	6µ/400n
TN1	6µ/400n
TN2	6µ/400n
TN3	1.2µ/200n
TN <sub>4</sub>	1.2µ/200n
TN5	1.2µ/200n
TN6	1.2µ/200n
TP2	2µ/400n
TP3	2µ/400n
TP <sub>4</sub>	2µ/400n
TP5	2µ/400n

 TABLE 2.2: Design parameters for the Latched Comparator

converter (ADC). Thus designing an comparator with less offset, meta-stability and noise will help in designing a better performing ADC, which is covered in the next chapter.

#### Analog-to-Digital Converter (ADC) Overview

An analog-to-digital converter (ADC) is a device which is able to convert an analog continuous signal to discrete digital signal. By nature, analog signals are continuous in amplitude and continuous in time; for example, temperature, pressure, natural sound, etc. On the other hand, digital signals are discrete in amplitude and discrete in time. In the modern age, all the digital equipment, including personal computers, can only *understand* digital signals. The ideal digital signal can be represented with only two states - '1' and 'o'.



FIGURE 3.1: Analog to digital conversion through ADC

The analog-to-digital conversion process involves sampling and a quantization process which introduces some errors. Every ADC has a range of frequencies within which it can operate without any major distortion called ADC bandwidth. The practical bandwidth of ADC can be defined by its sampling rate. If the sampling rate is twice the amount of the signal bandwidth, that ADC is capable of regenerating the signal without any aliasing. Resolution, which is defined by the number of output levels an ADC can quantize it's analog level to, is related to dynamic range of ADC. Thus, for various applications, different resolutions and sampling rates are needed, which ultimately create the need for different types of analog-to-digital converters.

#### 3.1 ANALOG-TO-DIGITAL CONVERTER APPLICATIONS

Analog-to-digital converters are used when an analog signal needs to be converted to digital. Although all the signals in the real world are analog, easiness of storing, re-constructing, and processing digital signals have increased the demand for digital data exponentially. Based on the required speed, power, and resolution, the ADC type is selected.



FIGURE 3.2: ADC types and applications based on speed and resolution [11]

Main applications can be categorized into: precise industrial measurements, voiceband and audio, data acquisition, high-speed instrumentation, high-speed video, and software defined radio [11]. Most of these applications can be covered with the Sigma-Delta ( $\Sigma - \Delta$ ) ADC, Successive Approximation Register (SAR) ADC and Pipeline ADC. For fast applications, Flash ADCs are used. Thus these three type of ADCs along with Flash ADC, are the most popular ADC types for high resolution and high speed applications. Adding to these, Single Slope, Dual Slope or Integrating ADC, and Time-Interleaved ADC also have many important applications.

Industrial measurement and instrumentation need high resolution analog-to-digital converters. Typically 16 to 24 bits are necessary along with a couple of hundreds Hertz of sampling rate. Integrating ADCs, including dual slope, triple slope, etc. used to serve this purpose, were very popular. But the Sigma-Delta ( $\Sigma - \Delta$ ) ADC slowly took their place and is now the main ADC to serve the previously mentioned purposes [11].

Data acquisition systems in measurements and process control systems have many applications. SAR ADCs are the best suited ADC type for this kind of applications [11]; especially when multiple channels are needed for multiplexing purpose. SAR
ADCs pose low to medium converting speed, accuracy is medium to high, but area is very small which made it perfect for using with sensors, or any other equipment from where data can be directly provided.

Voice band and audio systems needed very high resolution and very accurate data conversion. Adding to that, they need proper signal filtering and have specific signal-to-noise ratio (SNR) requirement. In a modern solid state memory era, all the audios need to be processed and stored in digital devices and have small areas for data conversion. Sigma-Delta ADCs are a very popular choice for these applications. They have high resolution (16 to 24 bits), very high accuracy, and do not need large area, making them perfect for these kind of applications [11].

High speed applications like digital video and display electronics need ADCs capable of converting the analog signal to digital quickly. Flash converters are a very popular choice for this purpose as they are very fast in operation. But as the requirements for high resolution ADC increased, flash converters couldn't fulfill that demand. Pipeline and SAR ADCs are fast in operation and resolution is also high [114]. Thus, these two ADCs saw lots of usage in high speed video and display electronics applications.

For software defined radio (SDR) applications, sampling rate should be very high as the radio frequency, or bandwidth used for these kind of applications are very large. Typically from 70 MHz to a couple of gigahertz of signal frequencies are used in SDR. Pipeline ADCs pose very high sampling rate, which is perfect for these kind of applications and widely used in the industry.

Flash ADCs are the fastest ADCs available. They normally work in the gigahertz range of sampling rate. Thus they have applications in radar systems, wideband radio receivers, electronic testing equipment, optical communications links, etc. But due to high power consumption, they do not have many low power and remote applications. Another disadvantage of Flash ADCs are that they offer low resolutions.

# 3.2 DESIGN STUDY OF ANALOG-TO-DIGITAL CONVERTERS (ADCS)

This section presents a summary of selected articles related to ADC design published in IEEE Journal of Solid-State Circuits (JSSC) between 2010 to 2014. ADC designs are categorized based on the architecture, design year, technology, resolution, sampling rate, power, and area. Fig. 3.3 maps a number of ADC types reported in JSSC in each of the years from 2010 to 2014. Pipeline ADCs were published in large numbers



FIGURE 3.3: Number of different ADC architectures published between 2010-14

consistently. Successive approximation (SA) and Time-Interleaved (TI) ADCs were also popular during that time frame. Other types of ADCs only have 1-2 papers in each year.

# 3.2.1 Different Technologies and Architectures

Fig. 3.4 shows the number of ADCs designed with different technologies from 2010 to 2014 and published in JSSC. In 2010, the number of ADCs designed with 90nm and 180nm were large. In 2012, 40nm, 65nm, 90nm, 130nm, and 180nm technologies were equally popular. In 2011, 2013, and 2014, 65nm technologies dominated the ADC



FIGURE 3.4: Number of ADCs with different technologies (nm)

Fig. 3.5 shows the number of ADCs designed with different architectures (2010-14) and published in JSSC. Pipeline and SAR ADCs were the two major architectures reported mostly. After that, Time-Interleaved and Sigma-Delta ADCs were equally popular. Other ADC design types were randomly reported.



FIGURE 3.5: Number of ADCs with different Architectures

ADC resolution depends on the number of analog levels in conversion and not related to technology. It is expected, since it is a basic design requirement. A higher resolution will mean a more precise converter. Fig. 3.6 shows the ADC resolution with different



FIGURE 3.6: ADC Resolution (Bits) with different Technologies (nm)

technologies. It seems, 65nm and 180nm were used more for designing high resolution ADCs compared to other technologies. Fig. 3.7 shows the ADC resolution with



FIGURE 3.7: ADC Resolution (Bits) with different Architectures

different architectures. Pipelined and SAR architectures typically show the largest

number of bits used for resolution. High speed ADCs, like Flash and Sigma-Delta have low resolution designs. Because, with high speed architecture, it is difficult to design high resolution ADCs. The overall observation is that, ADCs with resolutions between 4 and 12 are most popular and ranges in many applications.

The sampling rate is a frequency at which analog signal is sampled while converting to digital. It is not directly related to the bandwidth. Sampling rate should be twice the Nyquist frequency of a signal in order to replicate the signal from its digital form. Fig. 3.8 shows different ADC designs sampling frequency with different architectures.



FIGURE 3.8: ADC Sampling RateResolution (Bits) with different Architectures

Time Interleaved (TI) and Flash ADC architectures have the highest speed capabilities. Thus these two types of ADC also have the higher sampling rate. Flash ADC is the fastest ADC and possesses the highest sampling rate. Pipeline and Folding architectures also show high sampling rate, as these architectures are also designed for high speed applications along with Flash and Time Interleaved (TI) ADCs. Successive Approximation (SA) ADCs are generally the slowest type. Delta Sigma ADCs can also have high speed designs. ADCs designed for high speed applications generally use small fabrication technologies which can provide inherent speed advantages.

## 3.2.3 Power Consumption and Area

ADCs used for high speed applications need more power compared to the slow applications. Fig. 3.9 shows a pie-chart where different technologies are mapped with their ADC average power consumption. Smaller technologies, like 28nm, 40nm, 45nm, and 65nm are normally used for high speed applications where power requirement is high. It is also reflecting in the pie-chart that smaller technologies have higher average power.



FIGURE 3.9: ADC Average Power (mW) with Technologies (nm)

High speed ADC architecture needs a high amount of power. Fig. 3.10 shows a pie-chart where different ADC architectures are mapped with their average power consumption. Time Interleaved Pipeline ADCs' average power is higher than any other ADC. Also, Flash, Pipeline, and Time Interleaved show high power consumptions. It is very much expected as these ADCs are designed for high speed applications, which require high power consumption. Pulse Position Modulation, SAR, Cyclic, Two Step, and Delta-Sigma ADCs show low power consumption as they are generally not designed for high speed applications.

ADC area is an important factor for certain applications. Remote and mobile applications look for ADCs with smaller area. Fig. 3.11 shows a pie-chart where different technologies are mapped with average ADC area, which also shows that area



FIGURE 3.10: ADC Average Power (mW) with different Architectures

is highly process dependent. The higher the technologies (nm), the higher the active area requirement. Fig. 3.12 shows a pie-chart where different ADC architectures are



FIGURE 3.11: ADC Average area  $(mm^2)$  with Technologies (nm)

mapped with average ADC area. Hybrid ADCs and Time Interleaved Pipeline ADCs need more active area than any other architectures. It is expected as Time Interleaved architecture needs large numbers of ADCs to design which need more ADC design area.



FIGURE 3.12: ADC Average area  $(mm^2)$  with different Architectures

# 3.2.4 DNL and INL

INL (Integral Nonlinearity) error and DNL (Differential Nonlinearity) error are measures of how much the input to output levels deviate from a straight line, measured in LSB. Fig. 3.13 shows the average DNL values with technology change. They do not



FIGURE 3.13: Average DNL values with Technologies (nm)

show significant technology dependencies. Nearly all values are less than 1. Fig. 3.14 shows the average INL values with technology change. INL values more than *1* were

shown during this survey. INL values showed more inconsistency than DNL values. DNL and INL value changes were also observed with different ADC architectures.



FIGURE 3.14: Average INL values with Technologies (nm)

Fig. 3.15 shows the average DNL value change with different ADC architectures. These show very little overall DNL differences. Pulse Position Modulation, Time Interleaved, and Interpolated Subranging type ADCs showed larger average DNL values compared to others, which showed fairly similar values. Fig. 3.16 shows the



FIGURE 3.15: Average DNL values with different Architectures

average INL value change with different ADC architectures. Time Interleaved ADCs showed much larger INL values than the rest. Other designs showed comparatively

lower INL values. Flash ADCs showed lower average DNL and average INL values, which is significant.



FIGURE 3.16: Average INL values with different Architectures

# 3.2.5 Signal-to-Noise and Distortion Ratio (SNDR)

SNDR (Signal to Noise and Distortion Ratio) is related to noise and the strength of the signal compared to noise and harmonics. These are important for accuracy of conversion, and ensuring ENOB is closer to the resolution. Fig. 3.17 shows the average



FIGURE 3.17: Average SNDR values with Technologies (nm)

SNDR values with technology change and it does not show much process dependency. Only 5000nm technology showed very low SNDR value, which is expected from a bigger technology. Fig. 3.18 shows the average SNDR values with different ADC



FIGURE 3.18: Average SNDR values with different Architectures

architectures. Incremental type ADCs showed the largest SNDR value. Cont-Time Delta-Sigma, Cont-Time MASH, Hybrid, and Delta-Sigma Pipeline architectures also showed reasonably good SNDR values which help noise minimization and better ENOB. Delay Line ADCs showed the smallest SNDR values which also impact ENOB harshly. Time Interleaved Flash, Flash, and Interpolated Subranging architectures also showed small SNDR values, decreasing ENOB values.

Nyquist sampling rate ( $f_{snyq}$ ) is calculated by dividing sampling rate ( $f_s$ ) with oversampling ratio (*OSR*). And energy per Nyquist sample can be expressed as *Power*/ $f_{snyq}$ . Thus, power is directly proportional to the Nyquist sampling rate. The frequency at which SNDR is measured is expressed as  $f_{in,hf}$ .

ADC designs from 1997 to 2015, published in the IEEE International Solid-State Circuits Conference (ISSCC) and the IEEE Symposium on VLSI Technology Conference were studied, and Energy per Nyquist sample values are plotted against SNDR values (Fig. 3.19). Figure-of-Merit (FOM) were also plotted for all these designs which showed as an envelope for the energy values against SNDR values. From the plot, it is



FIGURE 3.19: SNDR values against Energy per Nyquist sample [12]

clear that most of the ADCs SNDR values lie between 50dB to 85dB. Although a large group of ADCs showed their SNDR values between 25dB and 45dB. One interesting observation is, most of the ADC design from 2015 showed less energy demand with SNDR value changes. SNDR values from all these ADC designs from 1997 to 2015,



FIGURE 3.20: SNDR values against frequencies  $(f_{in,hf})$  [12]

published in the IEEE International Solid-State Circuits Conference (ISSCC) and the IEEE Symposium on VLSI Technology Conference were also plotted against  $f_{in,hf}$  (Fig. 3.20). Clock Jitter values were also plotted for all these designs which showed as an

envelope for the SNDR values against  $f_{in,hf}$  values .

From the plot, it is shown that, if plotted against  $f_{in,hf}$ , most of the ADCs SNDR values lie between 45dB to 9odB. Although some ADCs showed their SNDR values between 2odB and 4odB. Adding to that, a large number of ADCs designed in 2015, showed  $f_{in,hf}$  higher than average ADCs'  $f_{in,hf}$  values, which are from 1997 to 2014.

# 3.2.6 ADC performance summary (2010-2014)

Table –3.1 lists a brief summary of ADCs designed in 2010 that were published in JSSC. From the table, it is clear that the Pipeline ADC dominated in 2010. The 1st design with 0.0019mW has the lowest power consumption, the 8th design is the best in terms of ENOB, the 3rd design is the best for SNDR, and 3rd and 4th designs have good DNL and INL performances. The 8th design has the highest speed.

Table –3.2 lists a brief summary of ADCs designed in 2011 that were published in JSSC. From the table, it is observed that not any single type of ADC dominated in 2011. The 1st design with 0.0013mW has the lowest power consumption, the 1st design is also the best in terms of ENOB, and it is also the best for SNDR with 57.7dB. The 7th ADC design is the overall best for DNL and INL performances with 0.47 and 0.49, respectively. The 8th design has the highest speed.

Table -3.3 lists a brief summary of ADCs designed in 2012 that were published in JSSC. It is clear that Pipeline and SAR were reported more than other ADCs. The 1st design with  $0.053\mu$ W has the lowest power consumption, the 3rd design is the best in terms of ENOB with 9.83/10, the 5th design is the best for SNDR, and the 2nd design has the best DNL and INL performances with 0.25 and 0.38, respectively. The 12th design has the highest speed.

Table –3.4 lists a brief summary of ADCs designed in 2013 that were published in JSSC. From the table, it is clear that SAR ADC was more popular in 2013. The 1st design has the lowest power consumption, the 8th design is the best in terms of ENOB with 4.8/5, and the 2nd one is the best for SNDR with 63dB. The 1st ADC design is the overall best for DNL and INL performances with 0.36 and 0.44, respectively. The 9th design is the fastest one.

Table -3.5 lists a brief summary of ADCs designed in 2014 that were published in

JSSC. It is clear that Pipeline, SAR and Time Interleaved ADCs dominated in 2014. The 1st design has the lowest power consumption, the 3rd design is the best in terms of ENOB with 9.73/10, the 4th design is the best for SNDR, and the 1st design has the best DNL and INL performances with 0.09 and 0.22, respectively. The last design is the fastest.

SNDR	( <b>dB</b> )	54.4	50.5	64	58.2	58	61.6	31.6	36.5
INL	(/LSB)	2.24	0.63	0.72	0.7	0.9	0.72	0.8	0.73
DNL	(/LSB)	0.49	0.47	0.26	0.35	9.0	0.39	0.8	0.53
ADC Type	4	SAR	Cyclic	Pipeline	Pipeline	Two Step	IT	Pipeline	Folding
ENOB	(Bits)	8.75	8.1	10.3	9.4	9.34	6.6	5.0	ъ.8
Supply	(2)	1.3	1	1.8	1.8	1	1.8	1.1	1
Area	( <i>mm</i> <sup>2</sup> )	0.36	0.02	0.26	0.36	0.36	13.3	0.03	0.36
Power	(MM)	0.0019	6.9	18.4	6.6	9	594	2.6	<u>5</u> 0
Sampling Rate	(MS/s)	1	50	50	50	100	160	2200	2700
Resolution	(Bits)	10	6	12	10	10	11	9	9
Technology	(uu)	65	90	180	180	90	350	40	06

JSSC	
l in	
published	
2010	
l in	
gnec	
desi	
s S	
ğ	
of A	
Brief summary (	
ŝ	
LABLE	

SNDR	(qB)	50.6	55	59	20.4	30.1	27.5	28
INL	(/LSB)	1.55	1.5	2.5	0.8	1.6	0.49	0.7
DNL	(/LSB)	0.78	1	0.4	0.5	1.4	0.47	0.6
ADC Type	1	SAR	Pipeline	Pipeline	Delay Line	Pipeline	TI Flash	IL
ENOB	(Bits)	8.11	8.8	7.5	3.6	4.7	4.3	4.9
Supply	()	1.2	1.4	2.5	1.2	1	1.1	1.5
Area	( <i>mm</i> <sup>2</sup> )	0.32	0.26	0.88	0.011	0.042	0.44	1.47
Power	(mW)	0.55	12.2	105	2	318	81	435
Sampling Rate	(MS/s)	40	100	800	1200	2400	12000	16000
Resolution	(Bits)	10	10	12	4	8	гО	9
Technology	(uu)	130	90	40	65	65	65	65

JSSC
published in
in 2011
lesigned i
EADCs d
ief summary of
TABLE 3.2: BI

SNDR	( <b>dB</b> )	56.7	57.97	50.4	46.8	64.5	58.3	55	56.6	44.5	30.5	38	58
INL	(/TSB)	0.46	0.38	0.77	1.1	1.54	0.69	1.36	1.5	1.5	0.65	0.68	2.4
DNL	(/LSB)	0.61	0.25	0.55	1.4	0.82	0.66	0.38	0.3	1.6	0.75	0.27	0.5
ADC	lype	SAR	SAR	SAR	SAR	Pipeline	Pipeline	Pipeline	Pipeline	SAR	SAR	IL	Pipeline
ENOB	(Bits)	9.1	9.18	9.83	7.5	10.4	9.4	9.13	9.1	7.1	4.8	9	9.3
Supply	()	1	0.6	3.3	0.5	1.2	2.5	1	1.8	1.2	1	1	2.5
Area	( <i>mm</i> <sup>2</sup> )	0.191	0.082	0.011	0.011	0.36	4	0.19	0.42	0.028	0.014	0.3	0.4
Power	(MM)	0.000053	0.001	0.058	0.0012	2.95	72	2:37	40	4	6.08	40	500
Sampling	Kate (MS/s)	0.001	0.2	0.768	1.1	30	80	200	300	400	1250	2200	3000
Resolution	(Bits)	10	10	10	8	12	10	10	11	8	9	7	12
Technology	(um)	130	180	180	40	90	250	65	40	65	40	65	40

$\mathbf{O}$
õ
S
Ĺ
Ц
•
Ч
ē
Ч
.S
Ξ.
-9
2
рц
2
H
0
2
•=
5
ð
Ē
60
Ğ
Ð
r"
$\leq$
Ц
$\checkmark$
Ę
$\geq$
H
2
Ц
Ц
H
3
£
٠Ĕ
31
щ
•••
$\tilde{\mathbf{c}}$
÷
(.)
Е
Г
В
<
ΓA

SNDR	( <b>dB</b> )	50.1	63	55	54.6	55.4	52.4	39-3	30.7	30.5	52	31.2	34
INL	(/LSB)	0.44	1.9	0.57	0.93	1.65	2	0.91	0.47	1.7	9	0.74	0.4
DNL	(/LSB)	0.36	0.97	0.58	0.52	0.52	0.5	0.79	0.67	0.67	0.8	0.49	0.5
ADC Tvne	- J L-	SAR	SAR	SAR	TI SAR	Pipeline	Pipeline	SAR	Flash	TI SAR	Pipeline	Flash	Flash
ENOB	(Bits)	8.4	10.1	8.84	8.8	8.9	8.4	6.23	4.8	4.75	10	4.89	5.35
Supply	(V)	0.5	0.6	1	1	1	1.2	1	1	1.3	2.5	1.5	0.9
Area	( <i>mm</i> <sup>2</sup> )	0.013	0.076	0.212	0.104	0.36	0.225	0.0015	0.007	0.24	0.5	2.75	0.27
Power	(MM)	0.00000065	0.000097	0.000053	2.3	26.6	33	3.1	0.595	20.1	240	76	242
Sampling Rate	(MS/s)	0.02	0.04	ы	170	300	1000	1200	1250	1600	2100	4100	10300
Resolution	(Bits)	6	12	10	10	10	10	8	ſŪ	9	12	9	9
Technology	(um)	40	65	65	65	65	65	32	65	90	40	90	40

JSSC
blished in
n 2013 pu
lesigned i
of ADCs o
3rief summary 6
<b>TABLE 3.4:</b> 1

TABLE 3.5: Brief summary of ADCs designed in 2014 published in JSSC

SNDR	( <b>dB</b> )	60.5	57.2	67	76.9	69.3	56.85	64.4	52.2	32.8	48	69	51.4	44.4	29.19	29.4	30.7
INL	(/LSB)	0.22	0.89	1.8	0.6	0.79	0.7	1.89	1.8	0.8	1.8	1	1	1.5	0.65	20	0.42
DNL	(/LSB)	0.09	0.5	0.2	0.53	0.78	0.7	0.28	0.9	0.8	0.87	0.3	1	0.75	0.19	3.4	0.47
ADC Type	-	SAR	SAR	SAR	Pipeline	Pipeline	SAR	Pipeline	Pipeline	Sub ranging	Pipeline	Pipeline	TI SAR	IL	TI SAR	IL	IL
ENOB	(Bits)	9.73	9.2	10.8	12.5	11.2	9.15	10.44	8.4	5.16	7.7	11.2	8.2	7.1	4.03	4.6	5.49
Supply	$\mathbf{\hat{S}}$	0.6	0.6	1.25	1.6	1.3	1.1	1.1	1	1.1	1	3.3	1	1.4	1.1	1.2	0.0
Area	( <i>mm</i> <sup>2</sup> )	0.12	0.103	0.11	1.43	0.5	0.0656	0.282	0.18	0.044	0.1	0.526	0.78	1.35	0.33	0.23	0.25
Power	(MM)	0.00000058	0.00039	4.04	67.8	6.38	2.86	30.7	19	9.9	7.1	1200	18.9	120	79.1	162	69.5
Sampling Rate	(MS/s)	0.016	0.1	5 2	60	70	80	200	800	1000	1000	1000	1000	4000	10000	12800	20000
Resolution	(Bits)	10	10	12	14	11	10	12	10	9	6	14	10	8	9	ſŪ	9
Technology	(uu)	180	180	65	180	130	40	55	65	65	65	65	65	65	65	65	32

# 3.3 HIGH SPEED-HIGH RESOLUTION ANALOG-TO-DIGITAL CONVERTERS (ADCS)

Most of the high resolution ADCs are slow in conversion and most of the high speed ADCs feature low resolution. Flash ADCs are the fastest ADC available, but suffers from low resolution. On the other hand, Sigma-Delta ADCs have high resolution, but are very slow. SAR ADCs have moderate speed and resolution.



FIGURE 3.21: State-of-the-art analog-to-digital converters (ADCs)

# 3.3.1 Importance of High Speed-High Resolution ADCs

High speed ADCs have many important applications. Digital video and display electronics are two of the most widely used applications of high speed ADCs. Wideband communication systems, wireless backhaul, software defined radio (SDR), satellite applications, ultrasound and medical imaging, radar detection and sonar applications, testing, measurement and instrumentation, and wireless gigabit communications are some of the very important applications where high speed ADCs are necessary. On the other hand, high resolution and high accuracy ADCs have also seen a wide variety of applications. Precise industrial measurements, data acquisitions, voice band and audio systems, temperature sensors, weigh scale, pressure sensors, microcontrollers, and processors are some of the most important applications of high resolution ADCs. SAR ADCs can serve for both high resolution and medium speed applications, but they have their own disadvantages including meta-stability issue.

## 3.3.2 Why high resolution Flash ADCs are not common

Flash ADCs are the fastest ADCs, capable of converting the analog to digital data in very high sampling rate. This kind of ADC is suitable for high bandwidth and fast applications. But Flash ADCs are highly power consuming, pose low resolution, and expensive. Furthermore, it also adds lot of capacitances at the input. Interpolating and folding architectures can save some power in this architecture. An n-bit Flash ADC consists of  $(2^n - 1)$  comparators that are directly used to measure analog data and convert it to digital data. For a 4-bit Flash ADC, 15 comparators are needed. Analog signal is fed into 15 comparators, each of which compares the analog value with pre-defined discrete reference values.

 $(2^n - 1)$  comparators also have  $(2^n - 1)$  discrete digital outputs as each comparator produce either '1' or 'o' after the comparison. These discrete outputs are called *Thermometer codes*. Regenerative latch at the comparator output store the results. These thermometer codes then go through a priority encoder, which encodes these data into digital out bit stream. Since all the comparators compare together, it is also called a *Parallel ADC* and has the highest sampling rate. A 5-bit Flash ADC needs 31 precise comparators, which can draw lots of power. Also, designing and fabricating a large number of highly precise comparators is a very difficult task.

When the output of a comparator cannot decide or distinguish between '1' and 'o', then comparator output faces meta-stability. If meta-stability occurs, it results an incorrect thermometer code and finally incorrect digital output. Thus, due to large number of required comparators and high power consumption, high resolution flash converters are not common.

#### 3.3.3 Successive Approximation Register (SAR) ADC Meta-stability

Working principle and architecture of Successive Approximation Register (SAR) ADCs are quite opposite to Flash ADCs. While in Flash ADCs, many comparators are utilized at the same time, in SAR ADCs, only one comparator is used in many cycles in a successive fashion to make the data conversion. SAR ADCs have various



FIGURE 3.22: Typical 3-bit Flash ADC architecture [7]

applications in different sectors, as they pose medium to high resolutions, medium to high sampling rates [13], and the required area is relatively small.

Due to the balance in speed, resolution, area, and power consumption, SAR ADCs have found many applications in the area of multi-channel data acquisition, motor control, sensor interfaces, instrumentations, power factor monitoring, UPS, process control, and remote applications. SAR ADCs use binary search algorithm for data conversion which is more power efficient than Flash ADCs'. In a SAR ADC, a *sample* & *hold circuit* samples the input with a Nyquist rate frequency. This sampled signal level is then compared with the output of a digital-to-analog converter (DAC). This is the sequential binary search process which is initialized by setting an N-bit register to  $\frac{V_{ref}}{2}$  value. If the sampled signal level is greater than  $\frac{V_{ref}}{2}$ , then output MSB remains at '1'. but if this sampled signal level is smaller than  $\frac{V_{ref}}{2}$ , then the output MSB bit is changed to 'o'. This process repeats successively N-times to generate N-bits: from MSB to LSB. Thus, the resolution of this comparator is N-bits [13].

Meta-stability is a comparison issue and occurs in successive binary searching process.



FIGURE 3.23: Typical SAR ADC architecture [13]

Although it can happen in other ADCs, SAR ADCs are the most sensitive to metastability. Metastable state occurs, when the output cannot decide the state as '1' or 'o'. In a SAR ADC, when a metastable value is at the input of the SAR ADC algorithm, the sub-sequence steps lose their tracks. Since the time requirement is very strict, the SAR ADC conversion algorithm becomes irreversible [115].

# 3.4 CHALLENGES IN DESIGNING HIGH SPEED-HIGH RESOLUTION ADCS

High speed and high resolution analog-to-digital converters have many important applications. But design and implementation of these kind of ADCs are very challenging. There are many design considerations and difficulties. For example, noise, meta-stability, offset, power, ENOB, sampling rate, etc. Also, different ADC architectures have their own design challenges and restrictions.

According to the Nyquist theorem, to re-construct an analog signal from it's digital form, sampling rate should be higher than twice the bandwidth of the input signal. The faster the sampling speed in an analog to digital conversion process, the more accurate re-production of the analog signal is possible. But with high sampling rate, comes higher bandwidth requirement and power consumption, which make designing low power ADCs more challenging.

High resolution applications require high resolution ADCs. Resolution reflects how

accurate and close the digital data is, compared to its analog form. The higher the resolution, the digital form is closer to the analog form. High resolution applications like audio systems and precise instrumentations require high resolution converters, while temperature sensor and pressure sensors can work with low resolution converters.

Signal-to-noise ratio (SNR) requirements differ for various applications. Audio systems have low noise tolerance level, since it degrades the audio quality. On the other hand, temperature and pressure sensors can be more flexible in terms of noise tolerance. Since it's directly related to the resolution of the converter, signal to noise level ratio is very important.

Effective number of bits (ENOB) is an ADC performance metric. For example, due to noise and distortion, a good 12-bit ADC's ENOB can be 11.5, while for a badly affected ADC, it can be 8. Also, ENOB is a performance comparison metric for ADCs with different resolutions. For example, an 11-bit ADC with an ENOB of 10.5 is way better than a 14-bit ADC with an ENOB of 11.

Power consumption is a major concern when it comes to high speed ADC design. The more the sampling rate, the more power is needed. Thus, it's a trade-off between speed and power consumption. For low power applications, such as cellphone, high speed ADC design is a challenge.

# CHAPTER 4 Soft Comparators

Traditional comparators have one common feature —they produce 1-bit digital output: either high state '1' or low state 'o'. This is why these comparators can be called 1-bit analog-to-digital converter (ADC). Adding to that, the comparison decision time is usually pretty long. In this work, a unique comparator design is proposed that is capable of producing multiple bits.

# 4.1 SOFT AND HARD DECISION DECODING

Hard decision decoding takes a stream of binary values, for example from a threshold detector where each binary bit is either '1' or 'o'. Input signal is sampled and compared with a single threshold value. If the voltage value is greater than the threshold voltage level, the binary level is labeled as '1', but if it is less than the threshold level, the binary level is labeled as 'o'. In soft decision decoding, input signal is compared with different threshold levels to find the closest binary value. Thus the output is not only 'o' or '1', but different combinations of 'o' and '1' to represent the closest amplitude level.

For understanding soft and hard decision decoding, let's assume an even parity encoder which generates possible codewords of 000, 011, 101, and 110. Also assume the interesting transmitted message is '01' which after the parity encoder becomes '011'. Now this '011' is transmitted through a communication channel and due to noise at the output of that channel it becomes '001'. Through the transmission channel, '0' is transmitted as '0 volt' and '1' is transmitted as '1 volt' [14]. For hard decision decoding, the only threshold voltage is 0.5 volt.

But '001' is not a valid codeword according to the parity encoder. The decoder in this case compares this output with all the possible codewords and calculates the minimum *Hamming Distance* and makes a decision when it finds the minimum distance.



FIGURE 4.1: Hard decision decoding system setup [14]

All possible Codewords	Hard decision output	Hamming distance
000	001	1
011	001	1
101	001	1
110	001	3

FIGURE 4.2: Hard decision decoding [14]

In this case, the minimum *Hamming Distance* is '1' and for that it has three valid codewords. The decoder may choose any of these and thus the probability of picking the correct codewords is 1/3.

For soft decoding, different threshold voltage levels are used for comparison and decision making. In this case, the received information is compared with all the possible codewords and minimum *Euclidean Distance* is measured. The codeword that has the minimum Euclidean distance is taken as the correct output data [14].

In this case, three threshold voltage levels 0.7V, 0.4V and 0.2V are found as the instant voltage level of the signal. The output level is compared with all the possible codewords, Euclidean distances area measured and the minimum distance is found as 0.49 which represents the correct codeword '011'.

Thus soft decision decoder uses more information than the hard decision decoder to find the correct transmitted codewords [14].



FIGURE 4.3: Soft decision decoding system setup [14]

Valid codewords	Voltage levels at each sampling instant of received waveform	Euclidean distance calculation	Euclidean distance
000	0.2V 0.4V 0.7V	$(0-0.2)^2 + (0-0.4)^2 + (0-0.7)^2$	0.69
( 0V 0V 0V )			
011	0.2V 0.4V 0.7V	(0-0.2) <sup>2</sup> + (1-0.4) <sup>2</sup> + (1-0.7) <sup>2</sup>	0.49
( 0V 1V 1V )			
101	0.2V 0.4V 0.7V	$(1-0.2)^2 + (0-0.4)^2 + (1-0.7)^2$	0.89
( 1V 0V 1V )			
110	0.2V 0.4V 0.7V	$(1-0.2)^2 + (1-0.4)^2 + (0-0.7)^2$	1.49
(1V1V0V)			

FIGURE 4.4: Soft decision decoding [14]

# 4.2 SOFT COMPARATOR AND HARD COMPARATOR CONCEPT

All the traditional comparators take two inputs to compare and produce one single bit output - either high '1' or low 'o' which resembles hard decision decoding concept. Thus those comparators can be named *Hard Comparator*. In this thesis, a comparator design is proposed which resembles the soft decision decoding concept. It utilizes the comparator decision transition time and instead of one single bit output, it produces multiple bits expressing different states of the comparator process. Because of the similarities with soft decision decoding concept, this comparator is referred as *Soft*  *Comparator*. Thus, one comparator is capable of producing multiple bits digital output. Furthermore, traditional Flash ADCs need large numbers of comparators. For example, while designing 12-bit Flash ADC,  $(2^{12} - 1)$  precise comparators are needed which require lots of power. Mismatch among the transistors can introduce offset. Using the proposed comparator, it is possible to design high resolution Flash ADCs with a smaller number of comparators. For example, if the soft comparator is capable of producing 4-bits of output, a 5-bit Flash ADC will only need 4 soft comparators, which is certainly capable of bring down the power requirements.

# 4.3 SOFT COMPARATOR ARCHITECTURE

The soft comparator basic structure is simple and divided into a few submodules. The heart of this circuit is a back-to-back NOT gate based dynamic latched comparator which can make the comparison decision quickly.



FIGURE 4.5: Soft Comparator basic block diagram

After the comparator, an XOR gate and a buffer are used followed by a delay line which is made out of tristate NOT gates, a NOT gate and a buffer circuit. The delay line is comprised of 32 delay cells connected in series. Each of these delay cells consists of a tristate NOT gate, a NOT gate and a buffer circuit. After the delay line circuit, a memory circuit is used. The memory circuit is comprised of 32 D-latches connected in parallel. Each of these D-latches is comprised of two transmission gates, a NAND gate and two NOT gates. At the end of these sub-modules, there is a 32-to-5 encoder circuit which encodes the 32 digital output from the memory circuit into 5 bit output. Fig. 4.5 shows the soft comparator basic block diagram consisting of major sub-modules.

# 4.3.1 Soft Comparator schematic diagram

The soft comparator schematic and layout are designed with Cadence Virtuoso IBM 130nm CMOS technology. Fig. 4.6 shows the soft comparator schematic diagram comprising all the sub-modules.



FIGURE 4.6: Soft comparator schematic diagram

Among all these sub-modules, the major sub-modules are the dynamic latched comparator, 32 delay cells or the delay line circuit, 32 D-latches in parallel or the memory circuit, and a 32-to-5 simple encoder circuit. For this soft comparator, two different supply voltages are used. The dynamic comparator uses 800mV ( $V_{dd1}$ ) and the rest of the circuits use 1.2V ( $V_{dd}$ ). At the comparator INN input, 600mV is used as input common mode voltage. Reset voltage is 1.2V and digital pulse source is used as Enable (En) signal.



FIGURE 4.7: Soft Comparator output with INP = 610mV and INN = 600mV

This enable signal switches from  $V_1$ =oV to  $V_2$ = 1.2V, its period is 130ns, pulse width is 65ns, the rise and fall time is 15ps and zero delay time is used. Another digital pulse source is used as the Clock signal (CLK). This clock signal switches from  $V_1$ =oV to  $V_2$ = 1.2V, period is 40ns, pulse width is 20ns, the rise and fall time is 15ps and zero delay time is used. Fig. 4.7 shows a 6-bit soft comparator output, where INP=610mv and INN=600mV are used. CLK signal and Enable (EN) signal are *High*. Here, Sign Bit ='1' and 5-bit digital output is '10110', which represents AO4=1, AO3=0, AO3=1, AO2=1, and AO0=0.

# 4.3.2 Dynamic Latched Comparator

The dynamic latched comparator used in the soft comparator design, is proposed in one energy efficient SAR ADC design and it is a dynamic two stage comparator [10]. The first stage is a pre-amplifier that performs the amplification of the input signal difference. The second stage acts as another voltage amplifier and also as a positive feedback amplifier which helps to create rail-to-rail voltage as digital output. This design was first designed for IBM 180nm CMOS and then was redesigned in IBM 130nm CMOS technology. For designing and simulating this dynamic latched comparator, a clock signal with a period of 40ns and pulse width of 20ns is used. The clock switches from  $V_1$ =oV to  $V_2$ = 1.2V, and its rising and falling times are 15ps each with no delay. The supply voltage  $V_{dd1}$  was used 800mV for this comparator.



FIGURE 4.8: Dynamic Two-stage Latched Comparator

For inputs INN=600mv and INP=610mV, supply voltage of 800mV, a clock signal of 40ns period and  $V_1$ =0V to  $V_2$ = 1.2V, the output node voltages (OUTP and OUTN) during the comparison phase are observed.



FIGURE 4.9: Comparator nodes OUTP and OUTN during comparison

The common mode voltage is 600mV which is applied at the input INN. If the input at INP now increases from 600Mv to higher values, for example up to 1.2V, the transition, decision making time, or output settling time of the dynamic comparator will be shorter. Because of bigger voltage difference between the inputs, it is easier for the comparator to make a comparison decision compared to the case when both input voltage values are very close. Here, output OUTP settling time is defined as the 95% voltage value of the maximum or, final OUTP value, which in this case is 1.2V. Thus when OUTP reaches the 95% of its final voltage value, the comparator makes a



FIGURE 4.10: Simplified model of Comparator in Latch Mode [7]

decision - either OUTP is '1' and OUTN is 'o' or, vice versa. For this comparator, latch time constant is very important. The time constant of the latch when the comparator is in latch mode can be calculated by analyzing a simplified back-to-back NOT gate based circuit shown in Fig. 4.10 [7]. If the output voltages  $V_x$  and  $V_y$  of the inverters are very close at the start of latch mode, they will be in linear range. At that point, each of these inverters can be modeled as voltage controlled current sources driving a RC load (Fig. 4.11) where  $A_v$  is a low-frequency gain of each of the inverters [7].



FIGURE 4.11: Linearized model of track-and-latch stage in Latch Mode [7]

For this linearized model, it can be written:

$$\frac{A_V}{R_L}V_y = -C_L(\frac{dV_x}{dt}) - (\frac{V_x}{R_L})$$
(4.1)

$$\frac{A_V}{R_L}V_x = -C_L(\frac{dV_y}{dt}) - (\frac{V_y}{R_L})$$
(4.2)

Time constant,  $\tau = R_L C_L$  and by rearranging the above equations, we get:

$$\left(\frac{\tau}{A_V - 1}\right)\left(\frac{d\Delta V}{dt}\right) = \Delta V \tag{4.3}$$

Where,  $\Delta V = V_x - V_y$ , is the inverters output voltage difference. Equation 4.3 is a first-order differential equation with solution,

$$\Delta V = \Delta V_o e^{\frac{(A_V - 1)t}{\tau}} = \Delta V_o e^{kt}$$
(4.4)

In equation 4.4, *k* is a constant, *t* represents the settling time  $(t_{st})$  and  $\Delta V_o$  is the initial voltage difference between the inputs.  $\Delta V$  can be expressed as  $\Delta V_{logic}$ , which is the voltage difference between bit=1 and bit=0. Thus settling time is a decreasing function of input voltage difference:

$$t_{st} = \frac{1}{k} ln(\frac{\Delta V_{logic}}{\Delta V_o}) \tag{4.5}$$

If  $\Delta V_o$  is small, the settling time will be large, and can be even larger than the allowed time for the latch mode. That incident is referred to as *Metastability*. In other words, if

the initial value difference  $\Delta V_o$  is very small, the differential output voltage does not increase enough to be recognized as the correct logic value [7].

In this dynamic comparator, by keeping the INN at 600mV, the INP value has been increased from 600.001mV to 1100mV. The output OUTP settling time is then plotted against the input voltage difference. Here, it is observed that, after around 2.9ns the curve becomes *saturated*. Thus, even after increasing the common mode input voltage difference beyond 500mV, the settling time does not reduce noticeably.



FIGURE 4.12: Output Settling Time with Input Common Mode Voltage Difference

Transistor	(W/L) Ratio
T42	870n/1µ
T45	870n/1µ
T44	4.4µ/2µ
T43	4.4µ/2µ
T46	4.4µ/2µ
T47	3µ/1µ
T48	3µ/1µ
T49	3µ/1µ
T50	3µ/1µ
T52	870n/1µ
T51	870n/1µ
T56	870n/1µ
T55	870n/1µ

TABLE 4.1: Design parameters for the Latched Comparator

## 4.3.3 XOR Gate and Buffer Circuit

One XOR gate is used immediately after the dynamic latched comparator. The purpose of this XOR gate is to make sure that OUTP and OUTN signals becomes totally of opposite states. When the comparator makes a comparison decision, then OUTP



FIGURE 4.13: XOR gate designed with 4 two-input NAND gate

should be '1' (if INP is greater than INN) and OUTN should be 'o' or, OUTP should be 'o' (if INP is smaller than INN) and OUTN should be '1'. As both OUTP and OUTN are analog signals, to make their decision a perfect digital, this XOR gate is used. The XOR gate is designed with 4 two input NAND gates. For designing a NAND gate, two PMOS of size (W/L=480nm/130nm) and two NMOS of size (W/L=480nm/130nm) are used. Fig. 4.14 shows the XOR gate checks whether the dynamic comparator



FIGURE 4.14: XOR Gate checks Comparator Output settling

output settles. Although this XOR gate makes sure that the OUTP and OUTN signals become a perfect digital and produce a high output at its own output, the digital signal at the XOR output degrades due to load effect. To restore the signal to its perfect digital shape, one buffer is used immediately after the XOR gate output. The buffer circuit is designed with two back-to-back NOT gates connected in a series. For the NOT gate, one PMOS transistor of size (W/L=480nm/130nm) and one NMOS transistor of size (W/L=240nm/130nm) are used.

# 4.3.4 Delay Line Circuit

In this soft comparator design, comparator settling time has been utilized to get more than a single bit information at the comparator output. To meet this purpose, a delay line circuit has been used. As mentioned, when the comparator inputs are very close, the settling time is the largest. For example, when INN =600mV and INP=601mV are used, the comparator positive output OUTP settling time is about 6.4ns. Thus for very small common mode input voltage difference, 6.4ns is the time duration the comparator needs to finally make a stable decision. Also, over a 500mV common mode input voltage difference, the settling time becomes almost saturated. Even after increasing the common mode input voltage difference beyond that, the settling time doesn't become significantly smaller.



FIGURE 4.15: Big Delay Cell circuit

Thus the total settling time 6.4ns is divided into two parts - from zero to 3ns, when the settling time doesn't get affected by common mode input voltage difference alteration and from 3ns to 6.4ns, when settling time drastically changes with common mode input voltage differences. The first 3ns can be expressed by a *Big Delay Cell* which delays the signal by around 3ns. This *Big Delay Cell* is comprised of a big tristate NOT gate, a NOT gate and a Buffer circuit (Fig. 4.15). The buffer is a simple circuit consisting of two back-to-back NOT gates. The big tristate NOT gate is a special NOT gate with an *Enable* signal (Fig. 4.16). When the *Enable (En)* signal is high, MOSFETs To and T1 will be activated and the circuit will work as a regular NOT gate but with

a bigger signal delay. But when the *Enable* (*En*) signal is low, MOSFETs To and T1 will be disconnected putting the output at a floating state, which will hold on to its previous value. Thus big tristate NOT gate acts as a *delayed-controlled NOT gate*.



FIGURE 4.16: Big Tristate NOT Gate circuit

Transistor	(W/L) Ratio
T2	2µ/1.5µ
Tı	2µ/1.5µ
То	1µ/6µ
T3	1µ/6µ

TABLE 4.2: Design parameters for the Big Tristate NOT Gate

The second part of the total delay, 3.4ns is divided among 32 smaller delay cells, making each small delay cell's delay around 0.1ns. The 32 delay cells are collectively named as delay line circuit or 32 delay cells (Fig. 4.17).

This circuit is capable of delaying the input signal by around 3.4ns. It consists of four sub-modules - four 8 delay cells and an n-type MOSFET. Its drain and source are connected making it a capacitor built with MOSFET. This MOSFET works as a dummy load for the delay line circuit to improve matching (Fig. 4.18). The total delay line circuit is sub-divided among four 8 delay cells in order to design the layout


FIGURE 4.17: Delay Line Circuit symbol

easily. The n-type MOSFET (TNo) is used here as a dummy load with size  $W=1\mu m$  and L=500nm.



FIGURE 4.18: Delay Line Circuit schematic diagram

Each sub-module has 8 small Delay Cells connected in a series. All the delay cells' *Enable (En)* signals are connected together so that they are active at the same time and also can be deactivated at the same time. The overall 0.85ns signal delay can be achieved with each of these 8 delay cell circuits. Each delay cell has one output, making it an eight-output delay circuit.

Each smaller delay cell is called only *Delay Cell* and is capable of delaying the input digital signal by around 0.1ns. The schematic of this circuit is exactly same as the big delay cell, but with a smaller tristate NOT gate. It is comprised of a tristate NOT gate, a NOT gate and a buffer circuit (Fig. 4.20).

The buffer is a simple circuit consisting of two back-to-back NOT gates. The tristate NOT gate is a NOT gate with an *Enable* signal (Fig. 4.21). But in this case,



FIGURE 4.19: 8 Delay Cells connected together



FIGURE 4.20: Delay Cell circuit

the tristate NOT gate is much smaller than the big tristate NOT gate as the later one cannot delay signal as much as the big tristate NOT gate. Thus tristate NOT gate acts as a *delayed-controlled NOT gate* with smaller delay capability.



FIGURE 4.21: Tristate NOT Gate circuit

Transistor	(W/L) Ratio
T2	380n/140n
T1	380n/140n
То	900n/140n
T3	900n/140n

TABLE 4.3: Design parameters for the Tristate NOT Gate

### 4.3.5 Memory Circuit

After the final comparison decision, the latched comparator output becomes *perfectly digital* through the XOR gate and buffer circuit. When the XOR gate becomes high, the propagation of *start comparison* signal through the delay cells stops.



FIGURE 4.22: D-latch based Memory circuit

Each of these 32 delay cells have outputs that represent the different states of the comparison phase. These 32 outputs need to be stored for any instance so that it can be ended in the subsequent stage. For this purpose, 32 D-latch circuits are used in parallel fashion (Fig. 4.19). 32 delay cells' outputs work as 32 D-latches' inputs. All 32 D-latches *Reset* signals are connected and set as *high* in normal operation. All 32 D-latches *CLK* signals are tied together and use Buffer output as the clock signal.

When dynamic comparator is done with the comparison decision, the output of the buffer becomes *high* and at the same moment memory circuit should *stop* taking



FIGURE 4.23: 32 D-latch connected in parallel

the data at its input and holds onto the last value. Thus the buffer output goes to the memory circuit through a big inverter designed with PMOS size W/L= $4\mu$ m/130nm and NMOS size W/L= $2\mu$ m/130nm. This big inverter makes the signal a *clean strong signal*. This inverted buffer output signal which is now a low state signal goes to each of these 32 D-latches *stop comparison* signal (Fig. 4.23).

The D-latch circuit is comprised of a NOT gate, which inverts the incoming clock signal from the big inverter, two transmission gates, one NAND gate and another NOT gate at the end. In this D-latch circuit, when Clock is low, first transmission gate gets disconnected and the output holds on to the last comparison state. But when the Clock is high, the incoming data *D* can pass through the first transmission gate, but the second transmission gate gets disconnected. Thus data goes through the NAND gate and then through the end NOT gate and data *D* can go all the way to output (Fig. 4.24).



FIGURE 4.24: Schematic of the used D-latch circuit

### 4.3.6 Encoder Circuit

After getting all the 32 memory circuit outputs, these outputs need to be encoded into 5- bits. For this purpose, immediately after the memory circuit, a 32-to-5 encoder circuit is used. It simply takes all the 32 memory circuit outputs and encodes them into 5-bit digital output (Fig. 4.25).



FIGURE 4.25: 32-to-5 Encoder Circuit symbol

The encoder circuit is comprised of 32 XOR gates, five 16-bit OR gates and an n-type dummy MOSFET (To) of size W=240nm and L=130nm, in a diode connected fashion (Fig. 4.26). This MOSFET To is used to prevent X1 output to become a floating node as X1 is not used in the encoder internal connections. Except the 32nd XOR gate, all other 31 XOR gates share one common input with its immediate XOR gate input.



FIGURE 4.26: Schematic of 32-to-5 Encoder circuit

The 16-bit OR gates are connected with the XOR gates with following equations -

$$AO0 = X17 + X18 + X19 + X20 + X21 + X22 + X23 + X24$$
(4.6)

$$+X25 + X26 + X27 + X28 + X29 + X30 + X31 + X32$$

$$AO1 = X9 + X10 + X11 + X12 + X13 + X14 + X15 + X16$$

$$+ X25 + X26 + X27 + X28 + X29 + X30 + X31 + X32$$
(4.7)

$$AO2 = X5 + X6 + X7 + X8 + X13 + X14 + X15 + X16$$

$$+ X21 + X22 + X23 + X24 + X29 + X30 + X31 + X32$$
(4.8)

$$AO3 = X3 + X4 + X7 + X8 + X11 + X12 + X15 + X16$$

$$+ X19 + X20 + X23 + X24 + X27 + X28 + X31 + X32$$
(4.9)

$$AO4 = X2 + X4 + X6 + X8 + X10 + X12 + X14 + X16$$

$$+ X18 + X20 + X22 + X24 + X26 + X28 + X30 + X32$$
(4.10)

Each of these five 16-bit OR gate are designed with four 4-bit NOR gates and one 4-bit NAND gate (Fig. 4.27). All the four 4-bit NOR gates outputs are acting as four inputs of the 4-bit NAND gate.



FIGURE 4.27: Schematic of 16-bit OR gate

Each of the 4-bit NOT gates was designed with 4 PMOS (W/L= $1.92\mu/130n$ ) transistors and 4 NMOS (W/L=240n/130n) transistors. On the other hand, the 4-bit NAND gate was designed with 4 PMOS (W/L=480n/130n) transistors and 4 NMOS (W/L=960n/130n) transistors.

### 4.4 HOW SOFT COMPARATOR WORKS

The whole purpose of designing the soft comparator is to utilize the dynamic latched comparator's output settling time and extract multiple bits of information. The dynamic comparator works with 800mV while the rest of the soft comparator circuit works with 1.2V supply voltage.

One input of the dynamic comparator INN is held at 600mV which works as the reference voltage. Input analog signal is applied at the other input INP. The dynamic comparator compares INP and INN and produces decision outputs OUTP and OUTN.

If INP is bigger than INN, then OUTP will rise to high and OUTN will fall to zero. If INP is smaller than INN, then OUTN will rise to high and OUTP will fall to zero. Either way after the comparison, both OUTP and OUTN will be in different states.

These two OUTP and OUTN signals are then used as input for the subsequent module XOR gate. As mentioned earlier, this XOR makes sure that OUTP and OUTN signals are completely in different states and produces a high signal at its own output, which indicates the comparison is done. But due to transient behavior the XOR output is not a *clean* digital signal. To restore this signal to a perfect digital output, a buffer circuit is used after the XOR gate. The buffer circuit makes sure that the signal restores to its perfect digital shape.

After the buffer circuit, one big delay cell is used which needs *Enable* high signal to commence working and a 'start comparison' high signal. This 'start comparison' signal comes from the dynamic comparator input 'start comparison' signal. Thus when the dynamic comparator 'start comparison' signal becomes high, it starts the comparison process. At the same time the big delay cell starts *delaying* the clock signal which it takes as input.

After the big delay cell, which delays the clock signal by around 3ns, a delay line circuit is used which delays the output of the big delay cell by roughly 0.1ns per delay cell. This delay line circuit has 32 outputs that are the 32 inputs of the memory circuit which stores these 32 state values. When the buffer circuit output becomes *perfectly* high, that indicates comparison and decision making is done. Thus the buffer output works as the clock signal for the memory circuit. As soon as the buffer output becomes high, the memory circuit stops taking inputs and stores its last state values. These 32 state values represent the settling times. The memory circuit has a *Reset* signal which can clear the memory when necessary and make the memory circuit ready for taking the next comparison input values.

The 32 outputs of the memory circuit then go to the 32-to-5 encoder circuit. The encoder circuit encodes the 32 digital signal into 5-bit digital values and shows them at the 5 outputs - from AOo to AO4. Thus the soft comparator produces 5-bit digital output corresponding to the magnitude and one sign bit based on OUTP for the polarity, which add up to 6-bits instead of one single bit digital output.

#### 4.5 SOFT COMPARATOR NOISE ANALYSIS

In a flash ADC, to generate a digital output with *m* bits, we need  $2^m - 1$  hard comparators with different thresholds. The difference between two consecutive comparator threshold voltages is one least significant bit (LSB). Each of them produces one bit output. When the input voltage is greater than the threshold, the output is "1" and otherwise the output is "0".

Since the number of comparators grows exponentially with m, using a soft overlapping comparator with multiple output bits is extremely useful. If the input voltage is greater than the threshold, then the first bit is "1" and other bits determine how much is the difference between the input and the threshold voltage. Similarly, for smaller inputs, the first bit is a "0" and other bits indicate the difference.

By smartly choosing the threshold voltages we can reduce the number of comparators while designing ADCs. This section discusses the performance of soft overlapping comparators in the presence of noise. First we analyzed the performance of a single soft comparator. Then we extended the idea to multiple soft comparators.

#### 4.5.1 *Single Soft Comparator*

Fig. 4.28 shows a simple model for a comparator, where an equivalent noise voltage source is added with the ideal input voltage and it causes different measured times and digital outputs. For simplicity, we assume that comparator's offset is removed



FIGURE 4.28: Single Soft Comparator

during the calibration process. The result of the calibration process is a noise-free curve that depicts the decision making time (t) versus input voltage ( $V_{in}$ ). A sample

calibration curve is depicted in Fig. 4.29 for a single soft comparator.

The x-axis denotes the voltage difference between the applied input voltage and



FIGURE 4.29: Decision-making time for different common mode voltage difference in a single soft comparator

threshold voltage and the y-axis denotes the required decision making time for the comparator. In the presence of noise, for a given  $V_{in}$  the decision making time  $t_{measured}$  is a random process. Let's assume we would like to determine  $V_{in}$  by measuring  $t_{measured}$  and using the curve in Fig. 4.29. We denote the estimated input voltage by  $\hat{V}$  and

$$t_{measured} = f(\hat{V}). \tag{4.11}$$

Thus for comparator *C* the estimated output can be written as

$$\hat{V} = V_{in} + n, \tag{4.12}$$

where (n) is an Additive White Gaussian Noise (AWGN) with zero mean and variance  $\sigma^2$ . Furthermore, let us assume, there are k points on the  $V_{in}$  axis, i.e.  $V_1, V_2, \dots, V_k$ , corresponding to "00..01", "00..10",  $\dots$ , "111..11" in digital format. For an (m + 1)-bit soft comparator we have  $k = 2^m - 1$  different voltage points on the  $V_{in}$  axis that

determine the equivalent digital output. In the presence of noise, using the curve in Fig. 4.29 we can measure the corresponding random variable  $\hat{V}$ . Then using (4.12), we can calculate the probability that  $V_{in}$  is within the interval  $(V_1, V_2)$ . Similarly, for other intervals the probability can be generated. Finally, the interval with the highest probability specifies the estimated output. Now consider a fixed input voltage  $V_{in}$  is applied to an (m + 1)-bit soft comparator. In general, the probability that  $V_{in}$  is within the interval  $(V_i, V_{i+1})$  can be calculated as:

$$Pr\{V_{i} < V_{in} < V_{i+1}\} = Pr\{V_{i} < \hat{V} - n < V_{i+1}\}\$$
  
$$= Pr\{V_{i} - \hat{V} < -n < V_{i+1} - \hat{V}\}\$$
  
$$= \frac{1}{2}\left[erf\left(\frac{V_{i+1} - \hat{V}}{\sigma\sqrt{2}}\right) - erf\left(\frac{V_{i} - \hat{V}}{\sigma\sqrt{2}}\right)\right], \quad (4.13)$$

where, in (4.13) the erf(x) is the well-known *error function* and  $\sigma^2$  denotes the noise variance. From (equation 4.13), we understand that the variance of input noise voltage drastically reduces the probability of correct estimation. In other words, the performance of the soft comparator depends on the noise level. We investigate the effect of noise on the performance of a single soft comparator. In a 3-bit soft comparator, two bits show the magnitude and one bit shows the polarity. Thus the calibration curve is divided by 3 different voltage points  $V_1 = T$ ,  $V_2 = 2T$  and  $V_3 = 3T$ , where T denotes interval length. For simulation, we applied a fixed input voltage  $V_{in}$  in interval (0, T) under different noise levels. We increased the noise level from o up to 100 mV and calculated the probability based on observing  $\hat{V}$ ,  $V_{in}$  is correctly detected. We repeated this simulation for different interval length  $T \in \{20, 30, 40, 50\}$ . Simulation results are presented in Fig. 4.30. As we can see larger intervals have better performance in comparison with smaller intervals. Furthermore, we calculated the probability that  $V_{in}$  is wrongly detected in other intervals. We repeated the simulations for two interval lengths T = 20 and T = 40. The results are presented in Fig. 4.31



FIGURE 4.30: Probability of correct detection for different interval lengths.

#### 4.5.2 M Soft Comparators with same threshold

In the last section we saw the effect of noise on the output of a single soft comparator. In this section we analyze the effect of multiple observations on estimation of  $V_{in}$ , using M soft comparators. A simple test model is presented in Fig. 4.32. Assuming M independent noise voltages with mean zero and variance  $\sigma^2$ , we can reduce the error probability by increasing the number of observations. Let us denote M different estimations of  $V_{in}$  by  $\hat{V}_{j}$ ,  $j \in \{1, 2, \dots, M\}$ . Then, by averaging on all observations, the value of  $V_{in}$  can be obtained with no error when  $M \to \infty$ .

First, we consider the case for two comparators C1 and C2. Using simulation results, we demonstrated that by increasing the number of observations better estimation for  $V_{in}$  can be obtained. When two samples,  $\hat{V}_1 = V_{in} + n_1$  and  $\hat{V}_2 = V_{in} + n_2$ , are observed, the error probability can be reduced in comparison to the single observation. In this case the average observation can be written as

$$\hat{V} = \frac{\hat{V}_1 + \hat{V}_2}{2} = V_{in} + \frac{n_1 + n_2}{2}.$$
(4.14)



(B) Interval length T = 40mV.

FIGURE 4.31: Probability of each interval when fixed  $V_{in}$  is applied in interval [0, T]



FIGURE 4.32: M Soft Comparators

To see any improvement compared to the single observation, the  $\Pr\left\{\left|\frac{n_1+n_2}{2}\right| < |n_1|\right\}$  should be higher than 50 %. We can simplify this probability as

$$\Pr\left\{ \left| \frac{n_1 + n_2}{2} \right| < |n_1| \right\} = \Pr\left\{ -|n_1| < \frac{n_1 + n_2}{2} < |n_1| \right\}$$
$$= \Pr\left\{ -2|n_1| - n_1 < n_2 < 2|n_1| - n_1 \right\}$$

As depicted in Fig. 4.33, assuming  $n_1 > 0$ , the observation gain can be obtained when Pr  $\{-3n_1 < n_2 < n_1\} > 0.5$ .



FIGURE 4.33: Two observations with normal distribution.

This probability can be written based on the well-known function erf(x)

$$\Pr\left\{-3n_1 < n_2 < n_1\right\} = \frac{1}{2}\left[erf\left(\frac{n_1}{\sigma\sqrt{2}}\right) + erf\left(\frac{3n_1}{\sigma\sqrt{2}}\right)\right] \ge 0.5 \tag{4.15}$$

Solving equation (4.15), we obtain a valid range  $0 < n_1 < 0.360$ , which can occur with probability equal to  $\frac{1}{2}erf\left(\frac{0.360}{\sigma\sqrt{2}}\right) = 0.14$ . Thus by increasing the number of observations from 1 to 2 we have 14 % more chance for better estimation.

Obviously, we can make better estimations by increasing the number of observations and averaging them. In order to do that, we repeated similar simulations for a group of *M* comparators with the same threshold. We assumed *M* different independent noise with variance  $\sigma^2$  for each comparator. Fig. 4.32 depicts a simple test setup for *M* soft comparators with the same threshold voltage.

It's clear that, by scaling down a random variable by M, its variance is divided by  $M^2$ . However, by adding M independent normal random variables, the variance will be multiplied by M, so the overall variance will be divided by M [116]. Mathematically explained, we might consider a weighted sum of independent variables  $x_i, i \in \{1, 2, \dots, M\}$  with normal distribution  $(\mu_i, \sigma_i^2)$  and coefficient  $c_i$ 

$$y = \sum_{i=1}^{M} c_i x_i.$$
 (4.16)

Then, the variable *y* has a normal distribution with mean  $\mu_y$  and variance  $\sigma_y^2$  where :

$$\mu_y = \sum_{i=1}^M c_i \mu_i, \tag{4.17}$$

$$\sigma_y^2 = \sum_{i=1}^M c_i^2 \sigma_i^2.$$
 (4.18)

For the test setup presented in Fig. 4.32, all noise distributions have equal variance and all the coefficients are the same and equal to 1/M. Thus, the overall observation has a normal distribution with mean 0 and variance  $\sigma_y^2 = \frac{\sigma^2}{M}$ . Averaging multiple outputs is a straight forward way to find a more stable output. Two other useful ways also exist. The first way is based on majority rule. In this method, all of the outputs of comparators are counted and the more probable output is the output with the maximum counter index.

The second way and also the ideal one to find the most probable output is based on multiplication. The idea behind this method is based on variable node operation for non-binary LDPC codes [117]. We know that for (m + 1)-bit soft comparators we have  $2^m$  different intervals and each of these intervals represent one element in  $\alpha \in GF(2^m)$ . Again consider the test setup presented in Fig.4.32. Assume that  $P_i, i \in \{1, 2, \dots, M\}$  is the probability vector of each comparator  $C_i$ . This probability vector has  $q = 2^m$  elements related to the probability of each symbol in  $GF(2^m)$ , i.e,

$$P_i = [p_i(\alpha)], \ \alpha \in GF(2^m).$$
(4.19)

Furthermore, let us denote by  $P^* = [p^*(\alpha)]$ ,  $\alpha \in GF(2^m)$ , the final estimated probability vector which can be found as

$$p^*(\alpha) = \mu \prod_{i=1}^M p_i(\alpha), \quad \alpha \in GF(2^m),$$
(4.20)

where parameter  $\mu$  is a normalized factor in such a way that  $\sum_{\alpha \in GF(2^m)} p^*(\alpha) = 1$ . Finally, the estimation of applied input voltage is equal to the corresponding element in  $P^*$  which has maximum probability value or

$$\hat{V} = \max[p^*(\alpha) \mid \alpha \in GF(2^m)]. \tag{4.21}$$

The simulation results for 8 observations for multiplication methods are presented in Fig. 4.34a and Fig. 4.34b, respectively for interval length T = 20 and T = 40.



(B) Interval length T = 40mV.

FIGURE 4.34: Effect of multiple observations for multiplication method. A fixed  $V_{in}$  is applied in interval [0, T]

# CHAPTER 5 CIRCUIT LAYOUT AND EVALUATION

For implementing this soft comparator, IBM 130nm CMOS technology was chosen. The total available area was limited by 4mm x 4mm. Out of this available area, 2mm x 2mm area was utilized for this chip, which was sufficient for this design purpose. The overall chip layout was designed step by step - from comparator to encoder.

### 5.1 CIRCUIT LAYOUT PLAN

The final chip layout contains four soft comparators, which are connected in an overlapping fashion to give it an ADC structure along with a separate dynamic comparator. The internal soft comparator sub-module layout configuration plan is shown in Fig. 5.1.



FIGURE 5.1: Soft Comparator Layout configuration plan

### 5.2 SOFT COMPARATOR LAYOUT

The soft comparator layout was constructed by attaching different sub-modules, like a comparator, a delay line circuit, a memory circuit, and an encoder circuit, according to the schematic design. For overall layout design, 7 layers of metals were used.

### 5.2.1 Dynamic Comparator

Fig. 5.2 shows the layout of the Dynamic Latched Comparator. The design rule checking (DRC) was ran every time to make sure there was no error.



FIGURE 5.2: Dynamic Comparator Layout

# 5.2.2 XOR Gate and Buffer Circuit

According to the schematic design, after the comparator cell, there is one XOR gate and a Buffer circuit. Here, the XOR gate is designed by connecting four 2 input NAND gates (Fig. 5.3). There was not any other specific design challenges except keeping the DRC clean. After the XOR gate, there is a Buffer circuit.



FIGURE 5.3: XOR Gate Layout

This Buffer circuit is simply designed by connecting two NOT gates in a series (Fig. 5.4).



FIGURE 5.4: Buffer Circuit Layout

# 5.2.3 Big Delay Cell and Delay Line Circuit

Immediately after the Buffer circuit, one Big Delay Cell is introduced in a series. The Big Delay Cell is comprised of a Big Tristate NOT gate, one NOT gate, and one Buffer circuit connected serially (Fig. 5.5). The Big Tristate NOT gate occupied a lot of space in the layout platform and there was not any other specific design challenges except following the design rules.



FIGURE 5.5: Big Delay Cell Layout

Followed by the Big Delay Cell, a Delay Line Circuit is introduced. A Delay Line Circuit is designed by connecting 32 Delay Cells in a series, where their *Enable* signal nodes are tied together (Fig. 5.6).

The first Delay Cell's output works as the input of next Delay Cell, which is connected in a series and in this way the rest of the Delay Cells are connected. At the end of the Delay Cell Line, one dummy n-type MOSFET, whose source and drain are tied together, is connected. This MOSFET works as a dummy load. Designing this Delay



FIGURE 5.6: Delay Line Circuit (Top and Bottom portions are shown)

Line Circuit asked for more metal layers as only metal 1 could not provide all the wiring requirement without touching each other. Thus metal 2 and metal 3 layers are introduced here to make efficient wiring connection.

# 5.2.4 Memory Circuit

The memory circuit consists of 32 D-Latches. Each of these D-Latches is designed by connecting two Transmission gates, two NOT gates, and one two-input NAND gate (Fig. 5.7).



The Memory Circuit has 32 D-Latches connected in parallel, with their clock signals

FIGURE 5.7: D-Latch Circuit Layout

tied together which comes through a Big Inverter (Fig. 5.8).



FIGURE 5.8: Memory Circuit Layout (Top and Bottom portions are shown)

#### 5.2.5 Encoder Circuit

Layout designing for the 32-to-5 Encoder Circuit was the most challenging task. It consists of five 16-bit OR gates, 32 XOR gates, and an n-type MOSFET added in a diode-connected fashion. (Fig. 5.9). Out of 32 XOR gates, 31 XOR gates' output were connected to 5 16-bit OR gate in different combinations, which asked for intelligent wiring connections. This Encoder layout also took the most amount of layout spaces. Each of the five 16-bit OR gates is designed by connecting four 4-bit NOR gates and one 4-bit NAND gate. All four outputs from the four 4-bit NOR gates act as the four inputs of the 4-bit NAND gate (Fig. 5.9).



FIGURE 5.9: 32-to-5 Encoder Layout and 16-bit OR Gate Layout

# 5.2.6 Soft Comparator

All the sub-modules layouts are connected according to the schematic design to build the soft comparator layout. The comparator, XOR gate, and the Big Delay Cell are put on the top side. The Delay Line Circuit, Memory Circuit, or the 32 D-Latches in parallel, and the 32-to-5 Encoder Circuit are located at the bottom of the soft comparator layout (Fig. 5.10). All the internal wiring followed the schematic wiring. DRC checking and Layout Versus Schematic (LVS) checking were pretty challenging, but were found to be error free.

	a second s		Burner i	STURST
	9			
ل <u>شقار کی ا</u> ب				
COLUMN & COLUMN		ومدوار منجور المجمع ومحجوا	2	
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1				
	THE OWNER WATER	أغلب تعرط كالأ الترك		
الأشتار وتوقد				يتعثدا
		الأر ودكدا للر		
		الفرخ تتبرك المراجع		S. R. A
				<b>1</b>
Constraint in street				
		اغر فبتعتاهم		
21.21.21.2				
		Local Local Section		
		Bank R. Start Physic		
	The latter way in the latter	ألتج وجالعها لترج		
Charles and a				
		ing 2 marching		
				<b>1 1 1 1</b>
		had a statement		100000000
		ألتج وبالبداعية التر		Real Property lies
31-21-51-51-51				77.8.1
				وعقدوا
200 Cart				
Construction of the local division of the lo		المردورة مراحة المرك		Concession of
		ر و ال مراجع ال ال		
<b>3 1 1 1 1</b>				
11 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1				7 8 3
		ألقب وبطلابها لترج		nonen
				1.8
5 - 5 5 5		Construction of Construction		The Reyard
Call ( Desires and other states	States and States	للارتوبالاتالان		
ALC: NO. OF THE OWNER		المرابع المرابع محمد المرابع الم		
		المحد تحجا ليرتب المت		
ومعدامهم و				
Allerations				

FIGURE 5.10: Soft Comparator Layout

### 5.3 METAL FILLING AND CAPACITORS

Soft Comparator layout is copied four times and put at the center of the layout window along with a Dynamic Comparator layout. These are then connected in a fashion so that testing can be done easily. All the wirings are done using metal layers from metal 1 to metal 6. As the total chip area is 2mm by 2mm and even after placing four Soft Comparators and one Dynamic Comparator, there were still large amount of blank spaces available. In order to keep the metal density uniformly throughout the chip, metal filling was required. For metal filling, different sizes of rectangular metal shapes were placed throughout the blank areas and connected to the ground (Fig. 5.11).



For electro static discharge (ESD) protection, twenty dual metal-insulator-metal



capacitors (dualmimcap) were used (Fig. 5.12). They also worked as Vdd, Vcc, and GND bypass capacitors.

Capacitor	Values	No of Capacitor
C1	255.1129p F	6
C2	279.8893p F	3
C3	341.8302p F	3
C4	317.0538p F	3
C5	238.2299p F	2
C6	261.3662p F	1
C <sub>7</sub>	319.2071p F	1
C8	296.0708p F	1

TABLE 5.1:	Dualmimcap	Capacitor	values
------------	------------	-----------	--------



FIGURE 5.12: Dualmimcap Capacitors used in final chip layout

### 5.4 PAD FRAME CREATION

Pad frame is the interface between the main circuit and the outer world. Physical pads are the points where bond wiring is done during the chip packaging process. Each of the pads has its own output power and ground rail.



FIGURE 5.13: Analog Pad, Digital Input Pad, and Digital Output Pad

They also have a ESD protection circuit. For this Soft Comparator and the Dynamic Comparator circuit connections, digital input and output pads, analog pads, Vdd pads, Vcc pads, and GND pads were used. After connecting all the pads together, it creates the pad frame.



FIGURE 5.14: Vdd Pad, Vcc Pad, and GND Pad

The Analog pad, Digital input pad, and Digital output pad are shown in Fig. 5.13. The Analog pad doesn't need any receiving or level shifting circuitry. It only contains ESD protection circuitry. This is used both as an analog input and output pad. Digital pads have similar shapes, but different internal components. The power supply pads - Vdd and Vcc look similar, but the ground pad GND looks different than the power supply pads (Fig. 5.14).

Before the pad frame was created, a pin configuration was planned. There are 36 pins in this chip circuitry. Thus 36 pads were needed for wire bonding. The final pin configuration corresponds to the pin output numbers (Fig. 5.15).

			36	35	8 3	<b>8</b>	ŝ	32	5	7 8	8	29			
			Vdd1 – Ana In	Vdd - Vdd			A10 - Dig Out	A11 – Dig Out	412 - Die Out		A13 - DIG OUT	A14 – Dig Out			
1	En - Dig In													A20 – Dig Out	28
	EII - Dig III					_								A21 – Dig Out	27
2	INP – Ana In	C	1	C6		So	ft Co	omp	1		C11	c	12	A22 – Dig Out	26
3	INN1 – Ana In		$\frac{1}{2}$	67		5	ft Co	mn	,				14	A23 – Dig Out	25
4	INN2 – Ana In		-			<u> </u>		mp	-		<u> </u>	Ť		A24 – Dig Out	24
5	INN3 – Ana In	C	3	C8		So	ft Co	omp	3		C19		16	A30 – Dig Out	23
6	INN4 – Ana In	6	4	C9		So	ft Co	omp	4		C17	/ c	18	A31 – Dig Out	22
7	CLK – Dig In		5	C10		$\vdash$					C19	,	20	A32 – Dig Out	21
8	INP_C – Ana In	╎└	_			Ľ	yn C	omp	,			_			
9	INN_C – Ana In													ASS - Dig Out	20
		L_	_				_	_	_			_		A34 – Dig Out	19
		OUTN – Ana	Out	OUTP – Ana Out	CND – gnd	A44 – Dig Out	A43 – Dig Out	0	A42 – Dig Out	A41 – Dig Out	A40 – Dig Out	Vec - Vec	A39 - 4CC		
		Ş	1	11	12	13	14	-	15	16	17	5	9		

FIGURE 5.15: Pad Frame Pin Configuration

### 5.5 FINAL CHIP LAYOUT

After filling the blank spaces with metal, adding the capacitors, creating the pads and pad frame and connecting them with the internal circuitry, final chip layout was almost done. One image bevel was used around the final chip layout which showed the physical border of the chip. It was also situated within the chip layout area, which is 2mm by 2mm. The final chip layout is shown in Fig. 5.16.



FIGURE 5.16: Final Chip Layout

### 5.6 CHIP EVALUATION

The overlapping Soft Comparator chip is designed with IBM 130nm CMOS technology. It was designed in room temperature, which is 27°C. It uses the supply voltage of 800mV for the Dynamic Comparator (which is tunable). The rest of the circuit supply voltage is 1.2V. Clock frequency is 25 MHz and the output is a 6-bit digital data (5-bit for magnitude and 1-bit for sign). The chip is simulated over temperature range from o°C to 80°C. Corner Simulation is also done to test the chip performances in the worst cases. In this design, the dynamic latched comparator uses 800mV supply voltage ( $V_{dd1}$ ), which is tunable. This supply voltage ( $V_{dd1}$ ) can be used for adjusting the accuracy in a calibration mode. Also, in the overall chip design, one single dynamic comparator is put together with 4 Soft Comparators. The positive output of that comparator (OUTP) can indicate whether the input is bigger or smaller than the reference signal. This is referred as *Sign Bit*. Thus the *Sign Bit* can verify the general functionality of the Soft Comparator.

### 5.6.1 *Performance test with Temperature variation*

The design has been simulated for a wide range of temperatures, starting from  $0^{\circ}$ C to  $80^{\circ}$ C with every  $5^{\circ}$ C difference. For INN = 600mV, INP = 625.6mV, the simulation ran for 12ns. Here, at a temperature of  $27^{\circ}$ C, for inputs INN = 600mV and INP = 625.6mV, the Soft Comparator Sign Bit produces '1' and the 5-bit binary output stream is '01000'. The decimal equivalent of that value is '8'.



FIGURE 5.17: Decimal Equivalent of Digital Output change with Temperature

But with temperature higher than the room temperature, the binary values start to decrease. On the other hand, for temperatures lower than the room temperature, the output binary values start to increase. Which means delaying in the delay cells decreases as temperature increases. The LSB value changes with temperature very rapidly, but the MSB value keeps it value almost constant over all temperature ranges. In the temperature ranges from 10°C to 50°C, the Soft Comparator binary value changes only '1' LSB bit.



FIGURE 5.18: Decimal Equivalent of Digital Output after  $V_{dd1}$  Calibration

By calibrating  $V_{dd1}$ , we can tune the digital output, so that for all the temperature from o°C to 80°C, the output remain the same (Fig. 5.18).

#### 5.6.2 *Corner Simulation*

The Soft Comparator chip is designed for *typical* behavior of both n-type and ptype MOSFETs, which is nom-nom (tt) option, supply Vdd of 1.2V and for room temperature of 27°C. The 'Corners' can be defined in various ways. The one used for this Soft Comparator, is shown in Table 5.3

The Soft Comparator is simulated to test the Corners. The simulation model is changed from 'tt' to 'ss', 'fs', 'sf' and, finally to 'ff'. For model, 'tt', p-type MOSFETs act as 'nom' and n-type MOSFETs also act as 'nom'. Under the conditions: temperature =  $27^{\circ}$ C,  $V_{dd}$  = 1.2V, INN = 600mV, and INP = 625.6mV, the Soft Comparator binary output bit stream was '01000' for this model.

Corners (x) & Variables (y)	slowslow (ss)	fastslow (fs)	typtyp (tt)	slowfast (sf)	fastfast (ff)
PMOS	slow	fast	nom	slow	fast
NMOS	slow	slow	nom	fast	fast

TABLE 5.2: Corner Definition used in Corner Simulation



FIGURE 5.19: Soft Comparator Output for typtyp (tt) Corners

For model, 'fs', p-type MOSFETs act as 'fast' and n-type MOSFETs act as 'slow'. Inputs were same: INN = 600mV and INP = 625.6mV. The Soft Comparator binary output bit stream for this model was '11111', which is way far from the original output '01000'.



FIGURE 5.20: Soft Comparator Output for fastslow (fs) Corners

For model, 'sf', p-type MOSFETs act as 'slow' and n-type MOSFETs act as 'fast'. Inputs were same: INN = 600mV and INP = 625.6mV. The Soft Comparator binary output bit stream for this model was '01001', which is only '1' LSB greater than the original output '01000'.

For model, 'ss', p-type MOSFETs act as 'slow' and n-type MOSFETs also act as 'slow'. Inputs were same: INN = 600mV and INP = 625.6mV. The Soft Comparator

.0400	125		
	64A		
<b>.</b> /401 d	0 22 2 0		
	2 R		
	1 8 1 1 9 10		
- 1112	10 10		
	7 8 5		
<b>-</b> 1403 - 4	125 10 17		
	8 1 n		
<b>-</b> ///04 •	32 2		
	21 00 A		
	101	40 13 30 <u>10 15 80</u>	

FIGURE 5.21: Soft Comparator Output for slowfast (sf) Corners

binary output bit stream for this model was '00111', which is only '3' bits smaller than the original output '01011'.



FIGURE 5.22: Soft Comparator Output for slowslow (ss) Corners

For model, 'ff', p-type MOSFETs act as 'fast' and n-type MOSFETs also act as 'fast'. Inputs were same: INN = 600mV and INP = 625.6mV. The Soft Comparator binary output bit stream for this model was '11111', which is way far from the original output '01000'.



FIGURE 5.23: Soft Comparator Output for fastfast (ff) Corners

Thus, it's been observed that the Soft Comparator performs fairly well in models 'tt' and 'sf'. It starts to change its value significantly in the 'ss' model. But the Soft Comparator could not perform close the 'tt' model, when it worked under the 'fs' and 'ff' models. Thus, the chip needs calibration and  $V_{dd1}$  can be used to calibrate the chip

Corners	V <sub>dd1</sub> (V)	5-bit Digital Output	Decimal Equivalent
typtyp (tt)	0.8	01000	8
fastslow (fs)	0.8	11111	31
slowfast (sf)	0.8	01001	9
slowslow (ss)	0.8	01011	11
fastfast (ff)	0.8	11111	31

TABLE 5.3: Corner Simulation Summary (without calibration)

to work in the 'fs', 'ss', and 'ff' models. Adding to that, signal processing modules can also be used in calibration. After the calibration, the performances of model 'ss' and 'sf' have been improved and matched the outcome with 'tt' model.

 TABLE 5.4: Corner Simulation Summary (after calibration)

Corners	<i>V<sub>dd1</sub></i> (V)	5-bit Digital Output	Decimal Equivalent
typtyp (tt)	0.8	01000	8
fastslow (fs)	-	11111	31
slowfast (sf)	0.81	01000	8
slowslow (ss)	0.825	01000	8
fastfast (ff)	-	11111	31

#### 5.6.3 Offset Calculation: Monte Carlo Simulation

In order to calculate the input referred offset voltage of the Soft Comparator, Monte Carlo Simulation was needed. A large number of sample points and a longer simulation time can give more accurate offset measurement. In this case, only 30 sample points were used and 2.4 $\mu$ s was used as the simulation time. The sampling method was chosen as *Random*. *Process* and *Mismatch* methods were taken into consideration. A common mode voltage of 600mV was used at both the inputs: INN and INP. A slowly rising *Ramp* voltage was applied at the INP input on top of the common

mode voltage of 600mV. The output of the Monte Carlo Simulation was observed and the longest transition time was located where the OUTP signal went through transition. It was found at the  $29^{th}$  sample point and at 960ns. The offset was calculated by multiplying the longest transition time with the slope of the *Ramp* signal.

$$Offset = (\frac{40mV}{2.4\mu s}) * 960ns = 16mV$$
(5.1)

FIGURE 5.24: Monte Carlo Simulation Output of the Soft Comparator

Fig. 5.24 is showing the Monte Carlo Simulation output of the Soft Comparator (time frame between 920ns and  $1.2\mu$ s are shown). The output is for 30 sample points and a simulation time of  $2.4\mu$ s. Fig. 5.25 is showing the longest transition incidents



FIGURE 5.25: Longest Transition at the 29<sup>th</sup> sample point in the simulation
in the Monte Carlo Simulation. The longest transition happened at the  $29^{th}$  sample point and it happened at 96ons.

Method	Process and Mismatch
Sampling Method	Random
Number of Points	30
Ramp Voltage	40mV
Simulation Time	2.4µs
Transition Sample	$29^{th}$
Transition Time	920ns
Offset	16mV

 TABLE 5.5: Monte Carlo Simulation Summary

## 5.6.4 *Performance Summary*

The performance of the Soft Comparator chip is summarized in Table 5.5.

 TABLE 5.6: Performance Summary of the Soft Comparator

Architecture	Soft Comparator
Technology	IBM 130nm CMOS
Resolution	6-bit (1 Sign
	Bit and 5-bit
	Digital Output)
Frequency	25 MHz
Supply V <sub>dd</sub>	1.2V
Supply V <sub>dd1</sub>	800mV (Tunable)
Power	26µ W

## CHAPTER 6

## CONCLUSION

Flash ADC, which is the fastest ADC around, cannot have high resolution due to power issue, meta-stability, lack of high number of accurate comparators, noise, and offset issues. Designing and fabricating a large number of similar and precise a comparator is very challenging. In this work, a different approach of designing a comparator is proposed where the comparator settling time is utilized to extract multiple bits of digital output instead of only one. This is referred to as a Soft Comparator. While output bits have different reliabilities, using a number of Soft Comparators together, it is possible to achieve high accuracy. Noise analysis of the Soft Comparator is done which showed that using multiple overlapping Soft Comparators, correct input detection is possible which is the pre-condition of getting reliable multiple bits output. As high resolution Flash ADCs need large number of precise traditional comparators, using the proposed Soft Comparator design, it is possible to design high resolution Flash ADCs with much fewer comparators. This reduces the power consumption drastically, alleviates meta-stability and offset error issues. Also, it can speed up the overall data conversion process.

## 6.1 FUTURE WORK

The fundamental concept behind a Soft Comparator is to utilize the dynamic comparator settling time to extract multiple bits of digital output instead of only one. This work used 32 small Delay Cells connected in series, each of which was capable of delaying the input signal by around 0.1ns. This Delay Cell was designed with one Tristate NOT gate, one NOT gate, and one Buffer circuit. These 32 Delay Cells together was named a 'Delay Line Circuit'. Thus, in this process, the Delay Line Circuit was comprised of 96 logic gates, which increases the overall power consumption.

In a future work, instead of these 32 individual delay cells, different lengths of metal wire can be used to create different delay times. For example, one short wire can

be connected at the output of Big Delay Cell, which will introduce a small amount of delay, a little longer wire can be connected at the output of the Big Delay Cell to get a little more amount of delay. This way, different lengths of wire are capable of delaying the same signal by different amounts of time. Thus, without 96 logic gates, power consumption could be much smaller and the overall process will be faster. Another approach is to use a long wire and connect it to the Big Delay Cell's output.

Then tapping it at different distances of the wire can give different amounts of delay. In this case, the process will be even faster, which is suitable for ultra-high speed data converter applications. Both of these alternating design approaches can be performed easily by removing the delay line circuit and replacing it with wiring delay systems. As the measured settling time with input voltage difference is non-linear, one Decision

Making Module (DMM) can be designed. Using Digital Signal Processing techniques inside the DMM, multiple bits reliable output can be generated.

In a future digitally assisted soft comparator design, the delay in the comparator can be modified by controlling its supply voltage  $V_{dd1}$  in a feedback loop to improve the accuracy of the comparator under different scenarios.

- [1] D. Cory, I. Hutchinson, and M. Chaniotakis. (2006) Operational Amplifier Circuits Comparators and Positive Feedback . [Online]. Available: http://ocw.mit.edu/courses/electrical-engineering-and-computer-science/ 6-071j-introduction-to-electronics-signals-and-measurement-spring-2006/ lecture-notes/24\_op\_amps3.pdf
- [2] H. Allen, P. E. Allen, and D. R. Holberg, CMOS Analog Circuit Design. N. MoPac Expy, Austin, TX: Holt, Rinehart and Winston, 1995.
- [3] I. ROHM Semiconductor. (2011) *Op-Amp/Comparator Tutorial*. [Online]. Available: http://rohmfs.rohm.com/en/products/databook/applinote/ic/ amp\_linear/common/opamp\_comparator\_tutorial\_appli-e.pdf
- [4] F. Larsen and M. Ismail, "A continuous-time offset compensated high speed BiCMOS comparator," in *Proceeding of IEEE International Symposium on Circuits and Systems (ISCAS).*, May 1996, pp. 285–288.
- [5] Y. Li and W. A. Serdijn, "A continuous-time level-crossing ADC with 1-bit DAC and 3-input comparator," in *Proceeding of IEEE International Symposium* on Circuits and Systems (ISCAS)., May 2012, pp. 1311–1314.
- [6] S. Naraghi, M. Courcy, and M. P. Flynn, "A 9-bit, 14μW and 0.06 mm<sup>2</sup> Pulse Position Modulation ADC in 90 nm Digital CMOS," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 9, pp. 1870–1880, Sept 2010.
- [7] T. C. Carusone, D. Johns, and K. Martin, *Analog Integrated Circuit Design*. River Street, Hoboken, N.J.: John Wiley & Sons, Inc, 2012.
- [8] L. Sandhya, "High speed latched comparator in 0.35 μ m CMOS process," in Proceeding of Global Conference on Communication Technologies (GCCT)., April 2015, pp. 150–154.
- [9] H. Jeon and Y. B. Kim, "A CMOS low-power low-offset and high-speed fully dynamic latched comparator," in *Proceeding of IEEE International SOC Conference* (SOCC)., Sept 2010, pp. 285–288.
- [10] M. van Elzakker, E. van Tuijl, P. Geraedts, D. Schinkel, E. A. M. Klumperink, and B. Nauta, "A 10-bit Charge-Redistribution ADC Consuming 1.9μW at 1 MS/s," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 5, pp. 1007–1015, May 2010.
- [11] W. Kester. (2005, Jun.) Which ADC Architecture Is Right for Your Application? [Online]. Available: http://www.analog.com/library/ analogDialogue/archives/39-06/architecture.html

- [12] B. Murmann. (2015) ADC Performance Survey 1997-2015 (ISSCC & VLSI Symposium). [Online]. Available: http://web.stanford.edu/~murmann/ adcsurvey.html
- [13] I. Ahmed, "Pipeline ADC Enhancement Technique," Ph.D. dissertation, Univ. Toronto, Toronto, ON, 2008.
- [14] M. Viswanathan, Simulation of Digital Communication Systems Using Matlab. Smashwords, 2013.
- [15] M. Chu, P. Jacob, J. W. Kim, M. R. LeRoy, R. P. Kraft, and J. F. McDonald, "A 40 gs/s Time Interleaved ADC using SiGe BiCMOS Technology," *IEEE Journal* of Solid-State Circuits, vol. 45, no. 2, pp. 380–390, Feb 2010.
- [16] C. T. Peach, U. K. Moon, and D. J. Allstot, "An 11.1 mW 42 MS/s 10 b ADC with Two-Step Settling in 0.18 μ m CMOS," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 2, pp. 391–400, Feb 2010.
- [17] Y. C. Huang and T. C. Lee, "A 0.02-mm<sup>2</sup> 9-bit 50-ms/s Cyclic ADC in 90-nm Digital CMOS Technology," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 3, pp. 610–619, March 2010.
- [18] Y. J. Kim, H. C. Choi, G. C. Ahn, and S. H. Lee, "A 12 bit 50 ms/s CMOS Nyquist A/D Converter With a Fully Differential Class-AB Switched Op-Amp," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 3, pp. 620–628, March 2010.
- [19] Y. Nakajima, A. Sakaguchi, T. Ohkido, N. Kato, T. Matsumoto, and M. Yotsuyanagi, "A Background Self-Calibrated 6b 2.7 GS/s ADC With Cascade-Calibrated Folding-Interpolating Architecture," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 4, pp. 707–718, April 2010.
- [20] O. Rajaee, T. Musah, N. Maghari, S. Takeuchi, M. Aniya, K. Hamashita, and U. K. Moon, "Design of a 79 dB 80 MHz 8X-OSR Hybrid Delta-Sigma/Pipelined ADC," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 4, pp. 719–730, April 2010.
- [21] K. Matsukawa, Y. Mitani, M. Takayama, K. Obata, S. Dosho, and A. Matsuzawa, "A Fifth-Order Continuous-Time Delta-Sigma Modulator With Single-Opamp Resonator," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 4, pp. 697–706, April 2010.
- [22] C. C. Liu, S. J. Chang, G. Y. Huang, and Y. Z. Lin, "A 10-bit 50-MS/s SAR ADC With a Monotonic Capacitor Switching Procedure," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 4, pp. 731–740, April 2010.
- [23] M. van Elzakker, E. van Tuijl, P. Geraedts, D. Schinkel, E. A. M. Klumperink, and B. Nauta, "A 10-bit Charge-Redistribution ADC Consuming 1.9  $\mu$  W at 1 MS/s," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 5, pp. 1007–1015, May 2010.

- [24] I. Ahmed, J. Mulder, and D. A. Johns, "A Low-Power Capacitive Charge Pump Based Pipelined ADC," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 5, pp. 1016–1027, May 2010.
- [25] A. Agah, K. Vleugels, P. B. Griffin, M. Ronaghi, J. D. Plummer, and B. A. Wooley, "A High-Resolution Low-Power Incremental *σ*- δ ADC With Extended Range for Biosensor Arrays," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 6, pp. 1099– 1110, June 2010.
- [26] Y. Zhu, C. H. Chan, U. F. Chio, S. W. Sin, S. P. U, R. P. Martins, and F. Maloberti, "A 10-bit 100-MS/s Reference-Free SAR ADC in 90 nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 6, pp. 1111–1121, June 2010.
- [27] J. Yang, T. L. Naing, and R. W. Brodersen, "A 1 GS/s 6 Bit 6.7 mw Successive Approximation ADC Using Asynchronous Processing," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 8, pp. 1469–1478, Aug 2010.
- [28] D. Gubbins, B. Lee, P. K. Hanumolu, and U. K. Moon, "Continuous-Time Input Pipeline ADCs," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 8, pp. 1456–1468, Aug 2010.
- [29] B. Verbruggen, J. Craninckx, M. Kuijk, P. Wambacq, and G. V. der Plas, "A 2.6 mW 6 bit 2.2 GS/s Fully Dynamic Pipeline ADC in 40 nm Digital CMOS," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 10, pp. 2080–2090, Oct 2010.
- [30] C. H. Law, P. J. Hurst, and S. H. Lewis, "A Four-Channel Time-Interleaved ADC With Digital Calibration of Interchannel Timing and Memory Errors," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 10, pp. 2091–2103, Oct 2010.
- [31] G. Taylor and I. Galton, "A Mostly-Digital Variable-Rate Continuous-Time Delta-Sigma Modulator ADC," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 12, pp. 2634–2646, Dec 2010.
- [32] B. Hershberg, S. Weaver, and U. K. Moon, "Design of a Split-CLS Pipelined ADC With Full Signal Swing Using an Accurate But Fractional Signal Swing Opamp," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 12, pp. 2623–2633, Dec 2010.
- [33] Y. H. Chung and J. T. Wu, "A CMOS 6-mW 10-bit 100-MS/s Two-Step ADC," IEEE Journal of Solid-State Circuits, vol. 45, no. 11, pp. 2217–2226, Nov 2010.
- [34] H. Marien, M. S. J. Steyaert, E. van Veenendaal, and P. Heremans, "A Fully Integrated  $\delta$ - $\sigma$  ADC in Organic Thin-Film Transistor Technology on Flexible Plastic Foil," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 1, pp. 276–284, Jan 2011.

- [35] S. K. Lee, S. J. Park, H. J. Park, and J. Y. Sim, "A 21 fJ/Conversion-Step 100 kS/s 10-bit ADC With a Low-Noise Time-Domain Comparator for Low-Power Sensor Interface," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 3, pp. 651–659, March 2011.
- [36] M. El-Chammas and B. Murmann, "A 12-GS/s 81-mW 5-bit Time-Interleaved Flash ADC With Background Timing Skew Calibration," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 4, pp. 838–847, April 2011.
- [37] C. C. Huang, C. Y. Wang, and J. T. Wu, "A CMOS 6-Bit 16-GS/s Time-Interleaved ADC Using Digital Background Calibration Techniques," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 4, pp. 848–858, April 2011.
- [38] M. Furuta, M. Nozawa, and T. Itakura, "A 10-bit, 40-MS/s, 1.21 mW Pipelined SAR ADC Using Single-Ended 1.5-bit/cycle Conversion Technique," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 6, pp. 1360–1370, June 2011.
- [39] P. J. A. Harpe, C. Zhou, Y. Bi, N. P. van der Meijs, X. Wang, K. Philips, G. Dolmans, and H. de Groot, "A 26 μ W 8 bit 10 MS/s Asynchronous SAR ADC for Low Energy Radios," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 7, pp. 1585–1595, July 2011.
- [40] T. Sundstrom, C. Svensson, and A. Alvandpour, "A 2.4 GS/s, Single-Channel, 31.3 dB SNDR at Nyquist, Pipeline ADC in 65 nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 7, pp. 1575–1584, July 2011.
- [41] E. Prefasi, S. Paton, and L. Hernandez, "A 7 mW 20 MHz BW Time-Encoding Oversampling Converter Implemented in a 0.08 mm<sup>2</sup> 65 nm CMOS Circuit," IEEE Journal of Solid-State Circuits, vol. 46, no. 7, pp. 1562–1574, July 2011.
- [42] S. H. Cho, C. K. Lee, J. K. Kwon, and S. T. Ryu, "A 550 μ W 10-b 40-MS/s SAR ADC With Multistep Addition-Only Digital Error Correction," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 8, pp. 1881–1892, Aug 2011.
- [43] P. Huang, S. Hsien, V. Lu, P. Wan, S. C. Lee, W. Liu, B. W. Chen, Y. P. Lee, W. T. Chen, T. Y. Yang, G. K. Ma, and Y. Chiu, "SHA-Less Pipelined ADC With In Situ Background Clock-Skew Calibration," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 8, pp. 1893–1903, Aug 2011.
- [44] Y. M. Tousi and E. Afshari, "A Miniature 2 mW 4 bit 1.2 GS/s Delay-Line-Based ADC in 65 nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 10, pp. 2312–2325, Oct 2011.
- [45] W. Liu, P. Huang, and Y. Chiu, "A 12-bit, 45-MS/s, 3-mW Redundant Successive-Approximation-Register Analog-to-Digital Converter With Digital Calibration," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 11, pp. 2661–2672, Nov 2011.

- [46] D. Vecchi, J. Mulder, F. M. L. van der Goes, J. R. Westra, E. Ayranci, C. M. Ward, J. Wan, and K. Bult, "An 800 MS/s Dual-Residue Pipeline ADC in 40 nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 12, pp. 2834–2844, Dec 2011.
- [47] P. J. A. Harpe, C. Zhou, K. Philips, and H. de Groot, "A 0.8-mW 5-bit 250-MS/s Time-Interleaved Asynchronous Digital Slope ADC," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 11, pp. 2450–2457, Nov 2011.
- [48] O. Rajaee, S. Takeuchi, M. Aniya, K. Hamashita, and U. K. Moon, "Low-OSR Over-Ranging Hybrid ADC Incorporating Noise-Shaped Two-Step Quantizer," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 11, pp. 2458–2468, Nov 2011.
- [49] M. Bolatkale, L. J. Breems, R. Rutten, and K. A. A. Makinwa, "A 4 GHz Continuous-Time δ-σ ADC With 70 dB DR and - 74 dBFS THD in 125 MHz BW," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 12, pp. 2857–2868, Dec 2011.
- [50] K. Doris, E. Janssen, C. Nani, A. Zanikopoulos, and G. van der Weide, "A 480 mW 2.6 GS/s 10b Time-Interleaved ADC With 48.5 dB SNDR up to Nyquist in 65 nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 12, pp. 2821–2833, Dec 2011.
- [51] K. A. O'Donoghue, P. J. Hurst, and S. H. Lewis, "A Digitally Corrected 5-mw 2-MS/s SC  $\delta$ - $\sigma$  ADC in 0.25  $\mu$  m CMOS With 94-dB SFDR," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 11, pp. 2673–2684, Nov 2011.
- [52] Y. H. Seo, J. S. Kim, H. J. Park, and J. Y. Sim, "A 1.25 ps Resolution 8b Cyclic TDC in 0.13  $\mu$  m CMOS," *IEEE Journal of Solid-State Circuits*, vol. 47, no. 3, pp. 736–743, March 2012.
- [53] S. Gupta, D. Gangopadhyay, H. Lakdawala, J. C. Rudell, and D. J. Allstot, "A 0.8-2 GHz Fully-Integrated QPLL-Timed Direct-RF-Sampling Bandpass σ δ ADC in 0.13 μ m CMOS," IEEE Journal of Solid-State Circuits, vol. 47, no. 5, pp. 1141–1153, May 2012.
- [54] C. Y. Chen, J. Wu, J. J. Hung, T. Li, W. Liu, and W. T. Shih, "A 12-Bit 3 GS/s Pipeline ADC With 0.4 *mm*<sup>2</sup> and 500 mW in 40 nm Digital CMOS," *IEEE Journal of Solid-State Circuits*, vol. 47, no. 4, pp. 1013–1021, April 2012.
- [55] A. Shikata, R. Sekimoto, T. Kuroda, and H. Ishikuro, "A 0.5 V 1.1 MS/sec 6.3 fJ/Conversion-Step SAR-ADC With Tri-Level Comparator in 40 nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 47, no. 4, pp. 1022–1030, April 2012.
- [56] C. J. Tseng, H. W. Chen, W. T. Shen, W. C. Cheng, and H. S. Chen, "A 10b 320-MS/s Stage-Gain-Error Self-Calibration Pipeline ADC," *IEEE Journal of Solid-State Circuits*, vol. 47, no. 6, pp. 1334–1343, June 2012.

- [57] P. J. A. Harpe, B. Busze, K. Philips, and H. de Groot, "A 0.47-1.6 mW 5-bit 0.5-1 GS/s Time-Interleaved SAR ADC for Low-Power UWB Radios," *IEEE Journal of Solid-State Circuits*, vol. 47, no. 7, pp. 1594–1602, July 2012.
- [58] D. Zhang, A. Bhide, and A. Alvandpour, "A 53-nW 9.1-ENOB 1-kS/s SAR ADC in 0.13- μ m CMOS for Medical Implant Devices," *IEEE Journal of Solid-State Circuits*, vol. 47, no. 7, pp. 1585–1593, July 2012.
- [59] J. K. R. Kim and B. Murmann, "A 12-b, 30-MS/s, 2.95-mW Pipelined ADC Using Single-Stage Class-AB Amplifiers and Deterministic Background Calibration," *IEEE Journal of Solid-State Circuits*, vol. 47, no. 9, pp. 2141–2151, Sept 2012.
- [60] I. N. Ku, Z. Xu, Y. C. Kuan, Y. H. Wang, and M. C. F. Chang, "A 40-mW 7bit 2.2-GS/s Time-Interleaved Subranging CMOS ADC for Low-Power Gigabit Wireless Communications," *IEEE Journal of Solid-State Circuits*, vol. 47, no. 8, pp. 1854–1865, Aug 2012.
- [61] S. Guhados, P. J. Hurst, and S. H. Lewis, "A Pipelined ADC With Metastability Error rate < 10<sup>-15</sup> Errors/Sample," *IEEE Journal of Solid-State Circuits*, vol. 47, no. 9, pp. 2119–2128, Sept 2012.
- [62] V. Singh, N. Krishnapura, S. Pavan, B. Vigraham, D. Behera, and N. Nigania, "A 16 MHz BW 75 dB DR CT  $\delta \sigma$  ADC Compensated for More Than One Cycle Excess Loop Delay," *IEEE Journal of Solid-State Circuits*, vol. 47, no. 8, pp. 1884– 1895, Aug 2012.
- [63] R. Xu, B. Liu, and J. Yuan, "Digitally Calibrated 768-kS/s 10-b Minimum-Size SAR ADC Array With Dithering," *IEEE Journal of Solid-State Circuits*, vol. 47, no. 9, pp. 2129–2140, Sept 2012.
- [64] T. Jiang, W. Liu, F. Y. Zhong, C. Zhong, K. Hu, and P. Y. Chiang, "A Single-Channel, 1.25-GS/s, 6-bit, 6.08-mW Asynchronous Successive-Approximation ADC With Improved Feedback Delay in 40-nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 47, no. 10, pp. 2444–2453, Oct 2012.
- [65] J. Guerber, H. Venkatram, M. Gande, A. Waters, and U. K. Moon, "A 10b Ternary SAR ADC With Quantization Time Information Utilization," *IEEE Journal of Solid-State Circuits*, vol. 47, no. 11, pp. 2604–2613, Nov 2012.
- [66] H. Wei, C. H. Chan, U. F. Chio, S. W. Sin, S. P. U, R. P. Martins, and F. Maloberti, "An 8-b 400-MS/s 2-b-Per-Cycle SAR ADC With Resistive DAC," *IEEE Journal* of Solid-State Circuits, vol. 47, no. 11, pp. 2763–2772, Nov 2012.
- [67] T. Miki, T. Morie, T. Ozeki, and S. Dosho, "An 11-b 300-MS/s Double-Sampling Pipelined ADC With On-Chip Digital Calibration for Memory Effects," *IEEE Journal of Solid-State Circuits*, vol. 47, no. 11, pp. 2773–2782, Nov 2012.

- [68] K. Reddy, S. Rao, R. Inti, B. Young, A. Elshazly, M. Talegaonkar, and P. K. Hanumolu, "A 16-mW 78-dB SNDR 10-MHz BW CT  $\delta \sigma$  ADC Using Residue-Cancelling VCO-Based Quantizer," *IEEE Journal of Solid-State Circuits*, vol. 47, no. 12, pp. 2916–2927, Dec 2012.
- [69] G. Y. Huang, S. J. Chang, C. C. Liu, and Y. Z. Lin, "A 1 μ W 10-bit 200-kS/s SAR ADC With a Bypass Window for Biomedical Applications," *IEEE Journal* of Solid-State Circuits, vol. 47, no. 11, pp. 2783–2795, Nov 2012.
- [70] Y. Chai and J. T. Wu, "A CMOS 5.37-mW 10-Bit 200-MS/s Dual-Path Pipelined ADC," IEEE Journal of Solid-State Circuits, vol. 47, no. 12, pp. 2905–2915, Dec 2012.
- [71] O. Rajaee and U. K. Moon, "Highly Linear Noise-Shaped Pipelined ADC Utilizing a Relaxed Accuracy Front-End," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 2, pp. 502–515, Feb 2013.
- [72] B. N. Fang and J. T. Wu, "A 10-Bit 300-MS/s Pipelined ADC With Digital Calibration and Digital Bias Generation," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 3, pp. 670–683, March 2013.
- [73] D. Stepanovic and B. Nikolic, "A 2.8 GS/s 44.6 mW Time-Interleaved ADC Achieving 50.9 dB SNDR and 3 dB Effective Resolution Bandwidth of 1.5 GHz in 65 nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 4, pp. 971–982, April 2013.
- [74] G. Taylor and I. Galton, "A Reconfigurable Mostly-Digital Delta-Sigma ADC With a Worst-Case FOM of 160 dB," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 4, pp. 983–995, April 2013.
- [75] B. D. Sahoo and B. Razavi, "A 10-b 1-GHz 33-mW CMOS ADC," IEEE Journal of Solid-State Circuits, vol. 48, no. 6, pp. 1442–1452, June 2013.
- [76] J. I. Kim, B. R. S. Sung, W. Kim, and S. T. Ryu, "A 6-b 4.1-GS/s Flash ADC With Time-Domain Latch Interpolation in 90-nm CMOS," IEEE Journal of Solid-State Circuits, vol. 48, no. 6, pp. 1429–1441, June 2013.
- [77] M. Yip and A. P. Chandrakasan, "A Resolution-Reconfigurable 5-to-10-Bit 0.4to-1 V Power Scalable SAR ADC for Sensor Applications," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 6, pp. 1453–1464, June 2013.
- [78] T. Oh, N. Maghari, and U. K. Moon, "A Second-Order  $\delta\sigma$  ADC Using Noise-Shaped Two-Step Integrating Quantizer," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 6, pp. 1465–1474, June 2013.
- [79] J. Wu, C. Y. Chen, T. Li, L. He, W. Liu, W. T. Shih, S. S. Tsai, B. Chen, C. S. Huang, B. J. J. Hung, H. T. Hung, S. Jaffe, L. K. Tan, and H. Vu, "A 240-mW 2.1-GS/s 52-dB SNDR Pipeline ADC Using MDAC Equalization," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 8, pp. 1818–1828, Aug 2013.

- [80] E. Z. Tabasy, A. Shafik, S. Huang, N. H. W. Yang, S. Hoyos, and S. Palermo, "A 6-b 1.6-GS/s ADC With Redundant Cycle One-Tap Embedded DFE in 90-nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 8, pp. 1885–1897, Aug 2013.
- [81] S. S. Wong, U. F. Chio, Y. Zhu, S. W. Sin, S. P. U, and R. P. Martins, "A 2.3 mW 10-bit 170 MS/s Two-Step Binary-Search Assisted Time-Interleaved SAR ADC," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 8, pp. 1783–1794, Aug 2013.
- [82] C. H. Chan, Y. Zhu, S. W. Sin, S. P. U, R. P. Martins, and F. Maloberti, "A 5-Bit 1.25-GS/s 4x-Capacitive-Folding Flash ADC in 65-nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 9, pp. 2154–2169, Sept 2013.
- [83] R. Sekimoto, A. Shikata, K. Yoshioka, T. Kuroda, and H. Ishikuro, "A 0.5-V 5.2-fJ/Conversion-Step Full Asynchronous SAR ADC With Leakage Power Reduction Down to 650 pW by Boosted Self-Power Gating in 40-nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 11, pp. 2628–2636, Nov 2013.
- [84] Y. Chae, K. Souri, and K. A. A. Makinwa, "A 6.3 μ W 20 bit Incremental Zoom-ADC with 6 ppm INL and 1 μ V Offset," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 12, pp. 3019–3027, Dec 2013.
- [85] P. Harpe, E. Cantatore, and A. van Roermund, "A 10b/12b 40 kS/s SAR ADC With Data-Driven Noise Reduction Achieving up to 10.1b ENOB at 2.2 fJ/Conversion-Step," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 12, pp. 3011– 3018, Dec 2013.
- [86] A. Varzaghani, A. Kasapi, D. N. Loizos, S. H. Paik, S. Verma, bookmark Command="[Quick Mark]"?>Sotirios, and S. Sidiropoulos, "A 10.3-GS/s, 6-Bit Flash ADC for 10G Ethernet Applications," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 12, pp. 3038–3048, Dec 2013.
- [87] L. Kull, T. Toifl, M. Schmatz, P. A. Francese, C. Menolfi, M. BrÃd'ndli, M. Kossel, T. Morf, T. M. Andersen, and Y. Leblebici, "A 3.1 mW 8b 1.2 GS/s Single-Channel Asynchronous SAR ADC With Alternate Comparators for Enhanced Speed in 32 nm Digital SOI CMOS," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 12, pp. 3049–3058, Dec 2013.
- [88] Y. Miyahara, M. Sano, K. Koyama, T. Suzuki, K. Hamashita, and B. S. Song, "A 14b 60 MS/s Pipelined ADC Adaptively Cancelling Opamp Gain and Nonlinearity," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 2, pp. 416–425, Feb 2014.
- [89] T. Oh, H. Venkatram, and U. K. Moon, "A Time-Based Pipelined ADC Using Both Voltage and Time Domain Information," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 4, pp. 961–971, April 2014.

- [90] K. Kim, W. Yu, and S. Cho, "A 9 bit, 1.12 ps Resolution 2.5 b/Stage Pipelined Time-to-Digital Converter in 65 nm CMOS Using Time-Register," *IEEE Journal* of Solid-State Circuits, vol. 49, no. 4, pp. 1007–1016, April 2014.
- [91] T. Danjo, M. Yoshioka, M. Isogai, M. Hoshino, and S. Tsukamoto, "A 6-bit, 1-GS/s, 9.9-mW, Interpolated Subranging ADC in 65-nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 3, pp. 673–682, March 2014.
- [92] S. H. W. Chiang, H. Sun, and B. Razavi, "A 10-Bit 800-MHz 19-mW CMOS ADC," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 4, pp. 935–949, April 2014.
- [93] S. C. Lee and Y. Chiu, "A 15-MHz Bandwidth 1-0 MASH  $\sigma \delta$  ADC With Nonlinear Memory Error Calibration Achieving 85-dBc SFDR," *IEEE Journal* of Solid-State Circuits, vol. 49, no. 3, pp. 695–707, March 2014.
- [94] B. Malki, T. Yamamoto, B. Verbruggen, P. Wambacq, and J. Craninckx, "A 70 dB DR 10 b 0-to-80 MS/s Current-Integrating SAR ADC With Adaptive Dynamic Range," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 5, pp. 1173–1183, May 2014.
- [95] S. Hashemi and B. Razavi, "A 7.1 mW 1 GS/s ADC With 48 dB SNDR at Nyquist Rate," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 8, pp. 1739–1750, Aug 2014.
- [96] H. Wei, P. Zhang, B. D. Sahoo, and B. Razavi, "An 8 Bit 4 GS/s 120 mW CMOS ADC," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 8, pp. 1751–1761, Aug 2014.
- [97] Y. Duan and E. Alon, "A 12.8 GS/s Time-Interleaved ADC With 25 GHz Effective Resolution Bandwidth and 4.6 ENOB," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 8, pp. 1725–1738, Aug 2014.
- [98] M. El-Chammas, X. Li, S. Kimura, K. Maclean, J. Hu, M. Weaver, M. Gindlesperger, S. Kaylor, R. Payne, C. K. Sestok, and W. Bright, "A 12 Bit 1.6 GS/s BiCMOS 2x2 Hierarchical Time-Interleaved Pipeline ADC," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 9, pp. 1876–1885, Sept 2014.
- [99] J. Jin, Y. Gao, and E. SAanchez-Sinencio, "An Energy-Efficient Time-Domain Asynchronous 2 b/Step SAR ADC With a Hybrid r-2R/C-3C DAC Structure," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 6, pp. 1383–1396, June 2014.
- [100] S. K. Shin, J. C. Rudell, D. C. Daly, C. E. MuAśoz, D. Y. Chang, K. Gulati, H. S. Lee, and M. Z. Straayer, "A 12 bit 200 MS/s Zero-Crossing-Based Pipelined ADC With Early Sub-ADC Decision and Output Residue Background Calibration," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 6, pp. 1366–1382, June 2014.
- [101] S. Zeller, C. Muenker, R. Weigel, and T. Ussmueller, "A 0.039  $mm^2$  Inverter-Based 1.82 mW 68.6 dB-SNDR 10 MHz-BW CT-  $\sigma \delta$  -ADC in 65 nm CMOS Using Power- and Area-Efficient Design Techniques," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 7, pp. 1548–1560, July 2014.

- [102] F. M. Yaul and A. P. Chandrakasan, "A 10 bit SAR ADC With Data-Dependent Energy Reduction Using LSB-First Successive Approximation," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 12, pp. 2825–2834, Dec 2014.
- [103] H. Nakane, R. Ujiie, T. Oshima, T. Yamamoto, K. Kimura, Y. Okuda, K. Tsuiji, and T. Matsuura, "A Fully Integrated SAR ADC Using Digital Correction Technique for Triple-Mode Mobile Transceiver," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 11, pp. 2503–2514, Nov 2014.
- [104] E. Z. Tabasy, A. Shafik, K. Lee, S. Hoyos, and S. Palermo, "A 6 bit 10 GS/s TI-SAR ADC With Low-Overhead Embedded FFE/DFE Equalization for Wireline Receiver Applications," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 11, pp. 2560–2574, Nov 2014.
- [105] A. M. A. Ali, H. Dinc, P. Bhoraskar, C. Dillon, S. Puckett, B. Gray, C. Speir, J. Lanford, J. Brunsilius, P. R. Derounian, B. Jeffries, U. Mehta, M. McShea, and R. Stop, "A 14 Bit 1 GS/s RF Sampling Pipelined ADC With Background Calibration," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 12, pp. 2857–2867, Dec 2014.
- [106] F. van der Goes, C. M. Ward, S. Astgimath, H. Yan, J. Riley, Z. Zeng, J. Mulder, S. Wang, and K. Bult, "A 1.5 mW 68 dB SNDR 80 Ms/s 2 x Interleaved Pipelined SAR ADC in 28 nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 12, pp. 2835–2845, Dec 2014.
- [107] S. Lee, A. P. Chandrakasan, and H. S. Lee, "A 1 GS/s 10b 18.9 mW Time-Interleaved SAR ADC With Background Timing Skew Calibration," *IEEE Journal* of Solid-State Circuits, vol. 49, no. 12, pp. 2846–2856, Dec 2014.
- [108] V. H. C. Chen and L. Pileggi, "A 69.5 mW 20 GS/s 6b Time-Interleaved ADC With Embedded Time-to-Digital Calibration in 32 nm CMOS SOI," *IEEE Journal* of Solid-State Circuits, vol. 49, no. 12, pp. 2891–2901, Dec 2014.
- [109] Y. Dong, W. Yang, R. Schreier, A. Sheikholeslami, and S. Korrapati, "A Continuous-Time o-3 MASH ADC Achieving 88 dB DR With 53 MHz BW in 28 nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 12, pp. 2868–2877, Dec 2014.
- [110] H. K. Hong, W. Kim, H. W. Kang, S. J. Park, M. Choi, H. J. Park, and S. T. Ryu,
   "A Decision-Error-Tolerant 45 nm CMOS 7b 1 GS/s Nonbinary 2b/Cycle SAR ADC," *IEEE Journal of Solid-State Circuits*, vol. 50, no. 2, pp. 543–555, Feb 2015.
- [111] X. Xing and G. G. E. Gielen, "A 42 fJ/Step-FoM Two-Step VCO-Based Delta-Sigma ADC in 40 nm CMOS," IEEE Journal of Solid-State Circuits, vol. 50, no. 3, pp. 714–723, March 2015.
- [112] Y. Zhou, B. Xu, and Y. Chiu, "A 12 bit 160 MS/s Two-Step SAR ADC With Background Bit-Weight Calibration Using a Time-Domain Proximity Detector," *IEEE Journal of Solid-State Circuits*, vol. 50, no. 4, pp. 920–931, April 2015.

- [113] C. H. Chen, Y. Zhang, T. He, P. Y. Chiang, and G. C. Temes, "A Micro-Power Two-Step Incremental Analog-to-Digital Converter," *IEEE Journal of Solid-State Circuits*, vol. 50, no. 8, pp. 1796–1808, Aug 2015.
- [114] I. Analog Devices. (2005, Jun.) Analog-Digital Conversion. [Online]. Available: http://www.analog.com/library/analogDialogue/archives/39-06/ Chapter%208%20Data%20Converter%20ApplicationsF.pdf
- [115] J.-e. Eklund and C. Svensson, "Metastability determines the noise in fast and accurate a/d converters," in *International Workshop on ADC modeling and testing*, *ser. ADC Modeling anf Testing*, ML Halttunen J., Daponte P., Ed. Helsinki. Citeseer, 1997, pp. 171–176.
- [116] A. Papoulis and S. U. Pillai, *Probability, random variables, and stochastic processes*. Tata McGraw-Hill Education, 2002.
- [117] D. Declercq, M. Fossorier, and E. Biglieri, *Channel Coding: Theory, Algorithms, and Applications: Academic Press Library in Mobile and Wireless Communications.* Academic Press, 2014.